SECURITIES AND EXCHANGE COMMISSION

FORM 8-K

Current report filing

Filing Date: **1998-03-10** | Period of Report: **1998-03-08** SEC Accession No. 0000893750-98-000087

(HTML Version on secdatabase.com)

FILER

LCI INTERNATIONAL INC /VA/

CIK:899760| IRS No.: 133498232 | State of Incorp.:DE | Fiscal Year End: 1231 Type: 8-K | Act: 34 | File No.: 001-12683 | Film No.: 98560729 SIC: 4813 Telephone communications (no radiotelephone) Mailing Address 4650 LAKEHURST COURT DUBLIN OH 43017

Business Address 8180 GREENSBORO DR STE 800 MCLEAN VA 22102 7034420220

UNITED STATES SECURITIES AND EXCHANGE COMMISSION WASHINGTON, D.C. 20549

Form 8-K

CURRENT REPORT

PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934

Date of Report (Date of earliest event reported) March 8, 1998

0-21602 (Commission File Number)

LCI INTERNATIONAL, INC.

(Exact name of Registrant as specified in its charter)

Delaware (State or incorporation) 13-3498232 (I.R.S. Employer Identification Number)

8180 Greensboro Drive, Suite 800, McLean, Virginia 22102 (Address of Registrant's principal executive office)

> 800-555-2124 (Registrant's telephone number)

ITEM 5. OTHER EVENTS

On March 8, 1998, LCI International, Inc., a Delaware corporation ("LCI"), entered into an Agreement and Plan of Merger (the "Merger Agreement") with Qwest Communications International Inc., a Delaware corporation ("Qwest"), and Qwest 1998-L Acquisition Corp., a wholly-owned subsidiary of Qwest ("Merger Sub"), providing for, among other things, the merger (the "Merger") of Merger Sub with and into LCI that will result in LCI becoming a wholly owned subsidiary of Qwest. The Merger Agreement and the joint press release of LCI and Qwest announcing the Merger are filed as Exhibit 2 and Exhibit 99, respectively, to this Form 8-K.

In connection with the execution of the Merger Agreement, on March 8, 1998, LCI, Philip F. Anschutz, the beneficial owner of not less than 170 million shares (the "Anschutz Shares") of Qwest common stock, and Anschutz Company, a Delaware corporation wholly owned by Mr. Anschutz and the record owner of the Anschutz Shares, entered into a Voting Agreement (the "Voting Agreement") providing for, among other things, (1) the obligation of Mr. Anschutz to cause Anschutz Company to vote the Anschutz Shares to approve the issuance of shares of Qwest common stock pursuant to the Merger Agreement and an amendment to Qwest's Amended and Restated Certificate of Incorporation increasing the number of authorized shares of Qwest common stock and (2) certain restrictions on (i) the sale or other transfer of the record ownership and the beneficial ownership, or both, of the Anschutz Shares by Mr. Anschutz or Anschutz Company and (ii) the acquisition by Mr. Anschutz or Anschutz Company of beneficial ownership of additional shares of capital stock of Qwest from any person other than Qwest, in each case until the consummation of the Merger or the termination of the Merger Agreement. The Anschutz Shares represent more than 80% of the issued and outstanding shares of Qwest common stock as of February 28, 1998. A copy of the Voting Agreement is attached as Exhibit 10.1 to this Form 8-K.

In connection with the execution of the Merger Agreement, on March 8, 1998, LCI entered into an amendment (the "Rights Amendment") to the Rights Agreement dated as of January 22, 1997 between LCI and Fifth Third Bank, as Rights Agent (the "Rights Agreement"), to the effect that Qwest, Merger Sub and their affiliates shall not become an Acquiring Person (as defined in the Rights Agreement) by reason of the execution of the Merger Agreement or the consummation of the Merger. A copy of the Rights Amendment is attached as Exhibit 10.2 to this Form 8-K.

The foregoing descriptions of the Merger Agreement, the Voting Agreement and the Rights Amendment and the transactions contemplated thereby do not purport to be complete and are qualified in their entirety by reference to the Merger Agreement, the Voting Agreement, and the Rights Amendment.

ITEM 7. FINANCIAL STATEMENTS, PRO FORMA FINANCIAL INFORMATION AND EXHIBITS

(c) Exhibits.

Exhibit Number

Title

- 2 Agreement and Plan of Merger dated as of March 8, 1998 among Qwest Communications International Inc., Qwest 1998-L Acquisition Corp. and LCI International, Inc.
- 10.1 Voting Agreement dated as of March 8, 1998 among Philip F. Anschutz, Anschutz Company and LCI International Inc.
- 10.2 First Amendment to Rights Agreement dated as of March 8, 1998 between LCI International, Inc. and Fifth Third Bank,

as Rights Agent.

Press Release of LCI and Qwest dated March 9, 1998

SIGNATURES

Pursuant to the requirements of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned, thereunto duly authorized.

LCI INTERNATIONAL, INC.

/s/ Lee M. Weiner By: -----Lee M. Weiner Vice President - General Counsel

Date: March 9, 1998

Exhibit Index

Exhibit Number

99

Title

- 2 Agreement and Plan of Merger dated as of March 8, 1998 among Qwest Communications International Inc., Qwest 1998-L Acquisition Corp. and LCI International, Inc.
- 10.1 Voting Agreement dated as of March 8, 1998 among Philip F. Anschutz, Anschutz Company and LCI International, Inc.
- 10.2 First Amendment to Rights Agreement dated as of March 8, 1998 between LCI International, Inc. and Fifth Third Bank, as Rights Agent.
 - 99 Press Release of LCI and Qwest dated March 9, 1998.

DATED AS OF MARCH 8, 1998

among

QWEST COMMUNICATIONS INTERNATIONAL INC.,

QWEST 1998-L ACQUISITION CORP.

and

LCI INTERNATIONAL, INC.

TABLE OF CONTENTS

ARTICLE I

	THE MERGER	2
1.1	The Merger	2
1.2	Closing	2
1.3	Effective Time	2
1.4	Effects of the Merger	2
1.5	Certificate of Incorporation	2
1.6	By-Laws	3
1.7	Officers and Directors of Surviving	
	Corporation	3
1.8	Effect on Capital Stock	3
1.9	Voting Agreement	4

ARTICLE II

	EXCHANGE OF CERTIFICATES 4
2.1	Exchange Fund
2.2	Exchange Procedures
2.3	Distributions with Respect to Unexchanged
2.4	Shares
- • 1	Stock \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots 5

Copyright © 2012 <u>www.secdatabase.com</u>. All Rights Reserved. Please Consider the Environment Before Printing This Document

2.5	No Fra	ctional Shares of Qwest Common Stock	. 5
2.6	Termin	ation of Exchange Fund	. 6
2.7	No Lia	bility	. 6
2.8	Invest	ment of the Exchange Fund	. 6
2.9	Lost C	ertificates	. 6
2.10	Withho	lding Rights	. 7
2.11	Furthe	r Assurances	. 7
2.12	Stock	Transfer Books	. 7
		ARTICLE III	
		ARTICLE III	
	REPRE	SENTATIONS AND WARRANTIES	. 8
3.1	Repres	entations and Warranties of LCI	. 8
	(a)	Organization, Standing and Power	
	(b)	Capital Structure	. 8
	(C)	Authority; No Conflicts	. 9
	(d)	Reports and Financial Statements	. 11
	(e)	Information Supplied	. 12
	(f)	Board Approval	. 12
	(g)	Vote Required	. 13
	(h)	Rights Agreement	. 13
	(i)	Brokers or Finders	. 13
	(j)	Opinion of LCI Financial Advisor	. 13
	(k)	Affiliate Letter	. 13
3.2	Repres	entations and Warranties of Qwest	. 13
	(a)	Organization, Standing and Power	. 13
	(b)	Capital Structure \ldots \ldots \ldots \ldots	. 14
	(C)	Authority; No Conflicts	. 15
	(d)	Reports and Financial Statements	. 16
	(e)	Information Supplied	. 17
	(f)	Board Approval	. 17
	(g)	Vote Required	
	(h)	Brokers or Finders	. 18
3.3	Repres	entations and Warranties of Qwest and	
	Merger	Sub	. 18
	(a)	Organization and Corporate Power	. 18
	(b)	Corporate Authorization	
	(C)	Non-Contravention	. 18
	(d)	No Business Activities	. 18
		ARTICLE IV	
	COVENANTS	RELATING TO CONDUCT OF BUSINESS	. 19
4.1	Covena	nts of LCI	. 19
	(a)	Ordinary Course	
	(b)	Dividends; Changes in Share Capital	
	(C)	Issuance of Securities	
	(d)	Governing Documents	
	(/	. ,	_ 5

Copyright © 2012 www.secdatabase.com. All Rights Reserved. Please Consider the Environment Before Printing This Document

(f) No Dispositions		
	• •	20
(g) Investments; Indebtedness		21
(h) Tax-Free Qualification		21
(i) Compensation		21
(j) Other Actions		21
(k) Accounting Methods; Income Tax		
Elections	•••	21
(1) Rights Agreement	•••	22
4.2 Covenants of Qwest	•••	22
(a) Ordinary Course	•••	22
(b) Dividends; Changes in Share Capital .	•••	22
(c) Issuance of Securities	•••	23
(d) Governing Documents	•••	23
(e) No Acquisitions	•••	23
(f) No Dispositions	•••	23
(g) Investments	•••	24
(h) Tax-Free Qualification	•••	24
(i) Other Actions	•••	24
(j) Accounting Methods; Income Tax		
Elections	•••	24
4.3 Advice of Changes; Governmental Filings	•••	24
4.4 Transition Planning	•••	25
4.5 Control of Other Party's Business	•••	25

ARTICLE V

5.

5. 5. 5.

5.

5. 5.

5. 5. 5. 5.

	ADDITIONAL AGREEMENTS 2	25
1	Preparation of Proxy Statement; LCI	
	Stockholders Meeting	25
2		26
3		27
4		27
5	Acquisition Proposals	29
6	Assumption of LCI Stock Options; Other Stock	
	Plans; Employee Benefits Matters 3	30
7	Fees and Expenses \ldots \ldots \ldots \ldots 3	31
8	Directors' and Officers' Indemnification and	
	Insurance	31
9		31
10		32
11		32
12	Listing of Shares of Qwest Common Stock 3	32
13	Voting Trust	32

ARTICLE VI

CONDITIONS	PRECEDENT						32

6.1	Conditions to Each Party's Obligation to	
	Effect the Merger	32
	(a) Stockholder Approval 3	33
	(b) No Injunctions or Restraints,	
	Illegality 3	33
	(c) FCC and Public Utility Commission	
	Approvals 3	33
	(d) HSR Act 3	33
	(e) NASDAQ Listing	33
	(f) Effectiveness of the Form S-4 3	33
6.2	Additional Conditions to Obligations of Qwest	
	and Merger Sub	33
	(a) Representations and Warranties 3	33
	(b) Performance of Obligations of LCI 3	33
	(c) Tax Opinion 3	34
6.3	Additional Conditions to Obligations of LCI 3	34
	(a) Representations and Warranties 3	34
	(b) Performance of Obligations of Qwest 3	34
	(c) Tax Opinion	34

ARTICLE VII

	TERMINATION AND AMENDMENT	35
7.1	Termination	35
7.2	Effect of Termination	36
7.3	Amendment	37
7.4	Extension; Waiver	37

ARTICLE VIII

	GENERAL PROVISIONS	38
8.1	Non-Survival of Representations, Warranties	
	and Agreements	38
8.2	Notices	38
8.3	Interpretation	39
8.4	Counterparts	39
8.5	Entire Agreement; No Third Party Beneficiaries	40
8.6	Governing Law	40
8.7	Severability	40
8.8	Assignment	40
8.9	Submission to Jurisdiction; Waivers	40
8.10	Enforcement	41
8.11	Definitions	41
8.12	Other Agreements	42

LIST OF EXHIBITS

Title

1.9	Form of Voting Agreement
5.6	Assumption of LCI Stock Options and LCI Warrant
6.2(c)(1)	Form of Qwest Tax Opinion
6.2(c)(2)	Form of Qwest Representations Letter
6.2(c)(3)	Form of LCI Representations Letter
6.3(c)(1)	Form of LCI Tax Opinion

GLOSSARY DEFINED TERMS

Definition

Location of Definition

Acquisition ProposalSection 5.5Adjustment ElectionSection 7.1(i)Adjustment Election PeriodSection 7.1(i)Affiliate AgreementSection 7.1(i)AgreementSection 7.1(i)Affiliate AgreementSection 7.1(i)AgreementSection 7.1(i)AgreementSection 7.1(i)AgreementSection 7.1(i)AgreementSection 7.1(i)AgreementPreambleAveragePriceSection 1.8(a)BenefitPlansSection 3.1(c) (iii)BlueSkyLaws
Adjustment Election Period
Affiliate Agreement<
Agreement<
Average Price
Benefit Plans
Blue Sky Laws
Board Approval
Board of Directors
Business Day
Certificate
Closing
Closing Date
Code
Communications Act
Confidentiality Agreement
Delaware Certificate of Merger
Determination Date
DGCL
DOJ
Effective Time
ERISA
Exchange Act
Exchange Agent
Exchange Fund
Exchange Ratio
-
Expenses
FCC
Form S-4
GAAP
Governmental Entity
HSR Act
LCI
LCI Affiliate Letter
LCI Benefit Plans
LCI Common Stock

LCI Disclosure Schedule
LCI Draft Disclosures
LCI Employees
LCI Evaluation Period
LCI Financial Advisor
LCI SEC Reports
LCI Stockholders Meeting
LCI Stock Option Plans
LCI Stock Options
-
LCI Transaction Information Section 3.1(d)(ii)
LCI Voting Debt
LCI Warrant
Material Adverse Effect
Merger
Merger Consideration
Merger Sub
NASDAQ
Person
Principal Stockholder
PUCs Section 3.1(c) (iii)
Quest
Qwest Charter Amendment
Qwest Common Stock
Qwest Disclosure Schedule
Qwest Draft Disclosures
Qwest Financial Advisor
Qwest SEC Reports
Qwest Stockholders Meeting
Qwest Transaction Information Section 3.2(d)(ii)
Qwest Voting Debt
Qwest Warrants
Regulatory Law
Required Consents
Required LCI Vote
Required Qwest Vote
Rights
Rights Agreement
Rule 145
SAS 72
SEC
Securities Act
Share Issuance
Subsidiary
Superior Proposal
Surviving Corporation
Termination Date
Termination Fee
Termination Notice
the other party

Violati	ion	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	Se	ect	io	n	3.1	(C)	(ii)
Voting	Aqr	ree	eme	ent	-																									S	ect	ion	1.	9

AGREEMENT AND PLAN OF MERGER

AGREEMENT AND PLAN OF MERGER, dated as of March 8, 1998 (this "Agreement"), among QWEST COMMUNICATIONS INTERNATIONAL INC., a Delaware corporation ("Qwest"), QWEST 1998-L ACQUISITION CORP., a Delaware corporation and a direct wholly-owned subsidiary of Qwest ("Merger Sub"), and LCI INTERNATIONAL, INC., a Delaware corporation ("LCI").

WITNESSETH:

WHEREAS, the respective Boards of Directors of Qwest, Merger Sub and LCI have each determined that the merger of Merger Sub with and into LCI (the "Merger") is in the best interests of their respective stockholders, and such Boards of Directors have approved such Merger, upon the terms and subject to the conditions set forth in this Agreement, pursuant to which each outstanding share of common stock, par value \$.01 per share, of LCI ("LCI Common Stock") issued and outstanding immediately prior to the Effective Time (as defined in Section 1.3), other than shares owned or held directly or indirectly by Qwest or directly by LCI will be converted into the right to receive shares of common stock, par value \$.01 per share, of Qwest ("Qwest Common Stock") as set forth in Section 1.8;

WHEREAS, Qwest, Merger Sub and LCI desire to make certain representations, warranties, covenants and agreements in connection with the transactions contemplated hereby and also to prescribe various conditions to the transactions contemplated hereby;

WHEREAS, Qwest, Merger Sub and LCI intend, by approving resolutions authorizing this Agreement, to adopt this Agreement as a plan of reorganization within the meaning of Section 368(a) of the Internal Revenue Code of 1986, as amended (the "Code"), and the regulations promulgated thereunder; and

WHEREAS, concurrently herewith the record and beneficial owners of not less than 80% of the outstanding shares of Qwest Common Stock (collectively, the "Principal Stockholder") has agreed to vote such shares in favor of the transactions contemplated hereby pursuant to the Voting Agreement (as defined in Section 1.9).

NOW, THEREFORE, in consideration of the foregoing and the respective representations, warranties, covenants and agreements set forth herein, and intending to be legally bound hereby, the parties hereto agree as follows:

ARTICLE I

THE MERGER

1.1 The Merger. Upon the terms and subject to the conditions set forth in this Agreement, and in accordance with the Delaware General Corporation Law (the "DGCL"), Merger Sub shall be merged with and into LCI at the Effective Time. Following the Merger, the separate corporate existence of Merger Sub shall cease and LCI shall continue as the surviving corporation (the "Surviving Corporation") under the name "LCI International, Inc."

1.2 Closing. The closing of the Merger (the "Closing") will take place on the second Business Day after the satisfaction or waiver (subject to applicable law) of the conditions (excluding conditions that, by their terms, cannot be satisfied until the Closing Date) set forth in Article VI (the "Closing Date"), provided, however, that, if the Average Price is less than \$26.9531, then the Closing shall not occur prior to (i) if LCI shall not deliver a Termination Notice (as defined in Section 7.1(i)) to Quest in accordance with Section 7.1(i), the second Business Day following the expiration of the LCI Evaluation Period (as defined in Section 7.1(i)) or (ii) if LCI shall deliver a Termination Notice to Qwest in accordance with Section 7.1(i), the second Business Day following the earlier of (A) Qwest's delivery of an Adjustment Election (as defined in Section 7.1(i)) and (B) the expiration of the Adjustment Election Period (as defined in Section 7.1(i)), in each case, unless another time or date is agreed to in writing by the parties hereto. The Closing shall be held at the offices of O'Melveny & Myers LLP, 153 East 53rd Street, New York, New York, 10022, unless another place is agreed to in writing by the parties hereto.

1.3 Effective Time. As soon as practicable following the Closing, the parties shall (i) file a certificate of merger (the "Delaware Certificate of Merger") in such form as is required by and executed in accordance with the relevant provisions of the DGCL and (ii) make all other filings or recordings required under the DGCL. The Merger shall become effective at such time as the Delaware Certificate of Merger is duly filed with the Delaware Secretary of State or at such subsequent time as Qwest and LCI shall agree and be specified in the Delaware Certificate of Merger (the date and time the Merger becomes effective being the "Effective Time").

1.4 Effects of the Merger. At and after the Effective Time, the Merger will have the effects set forth in the DGCL. Without limiting the generality of the foregoing, and subject thereto, at the Effective Time all the property, rights, privileges, powers and franchises of LCI and Merger Sub shall be vested in the Surviving Corporation, and all debts, liabilities and duties of LCI and Merger Sub shall become the debts, liabilities and duties of the Surviving Corporation.

1.5 Certificate of Incorporation. At the Effective Time, the certificate of incorporation of the Surviving Corporation shall be amended in accordance with the DGCL such that the certificate of incorporation of the Surviving Corporation shall consist of the provisions of the certificate of incorporation of Merger Sub, except that Article I of the Certificate of Incorporation of the Surviving Corporation shall be amended to read in its entirety as follows: "The name of this Corporation is 'LCI International, Inc.'".

1.6 By-Laws. The by-laws of Merger Sub as in effect at the Effective Time shall be the by-laws of the Surviving Corporation until thereafter changed or amended as provided therein or by applicable law.

1.7 Officers and Directors of Surviving Corporation. The officers of LCI as of the Effective Time shall be the officers of the Surviving Corporation, until the earlier of their resignation or removal or otherwise ceasing to be an officer or until their respective successors are duly elected and qualified, as the case may be. The directors of Merger Sub as of the Effective Time shall be the directors of the Surviving Corporation until the earlier of their resignation or removal or otherwise ceasing to be

a director or until their respective successors are duly elected and qualified.

1.8 Effect on Capital Stock.

At the Effective Time by virtue of the Merger and without (a) any action on the part of the holder thereof, each share of LCI Common Stock issued and outstanding immediately prior to the Effective Time (other than shares of LCI Common Stock owned by Qwest or Merger Sub or held by LCI, all of which shall be canceled as provided in Section 1.8(c)) shall be converted into the right to receive that number of shares of Qwest Common Stock equal to the Exchange Ratio (as defined below) (the "Merger Consideration"). "Exchange Ratio" means the quotient (rounded to the nearest 1/10,000) determined by dividing \$42 by the average (rounded to the nearest 1/10,000) (the "Average Price") of the volume weighted averages (rounded to the nearest 1/10,000) of the trading prices of Qwest Common Stock on the Nasdag National Market ("NASDAQ"), as reported by Bloomberg Financial Markets (or such other source as the parties shall agree in writing), for each of the 15 consecutive trading days ending on the trading day immediately preceding the date on which all the conditions to Closing (other than conditions that, by their terms, cannot be satisfied until the Closing Date) set forth in Article VI shall have been satisfied or waived (the "Determination Date"); provided, that the Exchange Ratio shall not be less than 1.0625 or, unless Qwest shall have exercised its rights under Section 7.1(i), greater than 1.5583.

(b) As a result of the Merger and without any action on the part of the holders thereof, at the Effective Time, all shares of LCI Common Stock shall cease to be outstanding and shall be canceled and retired and shall cease to exist, and each holder of a certificate which immediately prior to the Effective Time represented any such shares of LCI Common Stock (a "Certificate") (other than Merger Sub, Qwest and LCI) shall thereafter cease to have any rights with respect to such shares of LCI Common Stock, except the right to receive the applicable Merger Consideration in accordance with Article II upon the surrender of such certificate.

(c) Each share of LCI Common Stock issued and owned or held by

Qwest, Merger Sub or LCI at the Effective Time shall, by virtue of the Merger, cease to be outstanding and shall be canceled and retired and no stock of Qwest or other consideration shall be delivered in exchange therefor.

(d) Each share of common stock, par value \$.01 per share, of Merger Sub issued and outstanding immediately prior to the Effective Time, shall remain issued, outstanding and unchanged as validly issued, fully paid and nonassessable shares of common stock, par value \$.01 per share, of the Surviving Corporation as of the Effective Time.

1.9 Voting Agreement. Concurrently with the execution and delivery of this Agreement, LCI, and the Principal Stockholder are executing and delivering an agreement substantially in the form of Exhibit 1.9 hereto, pursuant to which, among other things, the Principal Stockholder is agreeing to vote all of the shares of Qwest Common Stock owned, beneficially or of record, by it to approve the Share Issuance and the Qwest Charter Amendment (as defined in Section 3.2(c)(i)).

ARTICLE II

EXCHANGE OF CERTIFICATES

2.1 Exchange Fund. Prior to the Effective Time, Qwest shall appoint Chase Mellon Shareholder Services or another commercial bank or trust company having net capital of not less than \$100,000,000, or a subsidiary thereof, to act as exchange agent hereunder for the purpose of exchanging Certificates for the Merger Consideration (the "Exchange Agent"). At or prior to the Effective Time, Qwest shall deposit with the Exchange Agent, in trust for the benefit of holders of shares of LCI Common Stock, certificates representing the Qwest Common Stock issuable pursuant to Section 1.8 in exchange for outstanding shares of LCI Common Stock. Qwest agrees to make available to the Exchange Agent from time to time as needed, cash sufficient to pay cash in lieu of fractional shares pursuant to Section 2.5 and any dividends and other distributions pursuant to Section 2.3. Any cash and certificates of Qwest Common Stock deposited with the Exchange Agent shall hereinafter be referred to as the "Exchange Fund".

2.2 Exchange Procedures. As soon as reasonably practicable after the Effective Time, the Surviving Corporation shall cause the Exchange Agent to mail to each holder of a Certificate (i) a letter of transmittal which shall specify that delivery shall be effected, and risk of loss and title to the Certificates shall pass, only upon delivery of the Certificates to the Exchange Agent, and which letter shall be in customary form and have such other provisions as Qwest may reasonably specify and (ii) instructions for effecting the surrender of such Certificates in exchange for the applicable Merger Consideration. Upon surrender of a Certificate to the Exchange Agent together with such letter of transmittal, duly executed and completed in accordance with the instructions thereto, and such other documents as may reasonably be required by the Exchange Agent, the holder of such Certificate shall be entitled to receive in exchange therefor (A) one or more shares of Qwest Common Stock representing, in the aggregate, the whole number of shares that such holder has the right to receive pursuant to Section 1.8 (after taking into account all shares of LCI Common Stock then held by such holder) and (B) a check in the amount equal to the cash that such holder has the right to receive pursuant to the provisions of this Article II, including cash in lieu of any fractional shares of Qwest Common Stock pursuant to Section 2.5. No interest will be paid or will accrue on any cash payable pursuant to Section 2.3 or Section 2.5. In the event of a transfer of ownership of LCI Common Stock which is not registered in the transfer records of LCI, one or more shares of Qwest Common Stock evidencing, in the aggregate, the proper number of shares of Qwest Common Stock, a check in the proper amount of cash in lieu of any fractional shares of Qwest Common Stock pursuant to Section 2.5 and any dividends or other distributions to which such holder is entitled pursuant to Section 2.3, may be issued with respect to such LCI Common Stock to such a transferee if the Certificate representing such shares of LCI Common Stock is presented to the Exchange Agent, accompanied by all documents required to evidence and effect such transfer and to evidence that any applicable stock transfer taxes have been paid.

2.3 Distributions with Respect to Unexchanged Shares. No dividends or other distributions declared or made with respect to shares of Qwest Common Stock with a record date after the Effective Time shall be paid to the holder of any unsurrendered Certificate with respect to the shares of Qwest Common Stock that such holder would be entitled to receive upon

surrender of such Certificate and no cash payment in lieu of fractional shares of Qwest Common Stock shall be paid to any such holder pursuant to Section 2.5 until such holder shall surrender such Certificate in accordance with Section 2.2. Subject to the effect of applicable laws, following surrender of any such Certificate, there shall be paid to such holder of shares of Qwest Common Stock issuable in exchange therefor, without interest, (a) promptly after the time of such surrender, the amount of any cash payable in lieu of fractional shares of Qwest Common Stock to which such holder is entitled pursuant to Section 2.5 and the amount of dividends or other distributions with a record date after the Effective Time theretofore paid with respect to such whole shares of Qwest Common Stock, and (b) at the appropriate payment date, the amount of dividends or other distributions with a record date after the but prior to such surrender and a payment date subsequent to such surrender payable with respect to such shares of Qwest Common Stock.

2.4 No Further Ownership Rights in LCI Common Stock. All shares of Qwest Common Stock issued and cash paid upon conversion of shares of LCI Common Stock in accordance with the terms of Article I and this Article II (including any cash paid pursuant to Section 2.3 or 2.5) shall be deemed to have been issued or paid in full satisfaction of all rights pertaining to the shares of LCI Common Stock.

2.5 No Fractional Shares of Qwest Common Stock.

(a) No certificates or scrip or shares of Qwest Common Stock representing fractional shares of Qwest Common Stock shall be issued upon the surrender for exchange of Certificates and such fractional share interests will not entitle the owner thereof to vote or to have any rights of a shareholder of Qwest or a holder of shares of Qwest Common Stock.

(b) Notwithstanding any other provision of this Agreement, each holder of shares of LCI Common Stock exchanged pursuant to the Merger who would otherwise have been entitled to receive a fraction of a share of Qwest Common Stock (after taking into account all Certificates delivered by such holder) shall receive, in lieu thereof, cash (without interest) in an amount equal to the product of (i) such fractional part of a share of Qwest Common Stock multiplied by (ii) the last sales price per share of Qwest Common Stock reported on NASDAQ in The Wall Street Journal, Eastern edition, with respect to the Closing Date. As promptly as practicable after the determination of the amount of cash, if any, to be paid to holders of fractional interests, the Exchange Agent shall so notify Qwest, and Qwest shall cause the Surviving Corporation to deposit such amount with the Exchange Agent and shall cause the Exchange Agent to forward payments to such holders of fractional interests subject to and in accordance with the terms hereof.

2.6 Termination of Exchange Fund. Any portion of the Exchange Fund which remains undistributed to the holders of Certificates for twelve months after the Effective Time shall be delivered to the Surviving Corporation or otherwise on the instruction of the Surviving Corporation, and any holders of the Certificates who have not theretofore complied with this Article II shall thereafter look only to the Surviving Corporation and Qwest for the Merger Consideration with respect to the shares of LCI Common Stock formerly represented thereby to which such holders are entitled pursuant to Section 1.8 and Section 2.2, any cash in lieu of fractional shares of Qwest Common Stock to which such holders are entitled pursuant to Section 2.5 and

any dividends or distributions with respect to shares of Qwest Common Stock to which such holders are entitled pursuant to Section 2.3. Any such portion of the Exchange Fund remaining unclaimed by holders of shares of LCI Common Stock five years after the Effective Time (or such earlier date immediately prior to such time as such amounts would otherwise escheat to or become property of any Governmental Entity (as defined in Section 3.1(c)(iii))) shall, to the extent permitted by law, become the property of the Surviving Corporation free and clear of any claims or interest of any Person previously entitled thereto.

2.7 No Liability. None of Qwest, Merger Sub, LCI, the Surviving Corporation or the Exchange Agent shall be liable to any Person in respect of any Merger Consideration from the Exchange Fund delivered to a public official pursuant to any applicable abandoned property, escheat or similar law.

2.8 Investment of the Exchange Fund. The Exchange Agent shall invest any cash included in the Exchange Fund as directed by the Surviving

Corporation on a daily basis. Any interest and other income resulting from such investments shall promptly be paid to the Surviving Corporation.

2.9 Lost Certificates. If any Certificate shall have been lost, stolen or destroyed, upon the making of an affidavit of that fact by the Person claiming such Certificate to be lost, stolen or destroyed and, if required by the Surviving Corporation, the posting by such Person of a bond in such reasonable amount as the Surviving Corporation may direct as indemnity against any claim that may be made against it with respect to such Certificate, the Exchange Agent will deliver in exchange for such lost, stolen or destroyed Certificate the applicable Merger Consideration with respect to the shares of LCI Common Stock formerly represented thereby, any cash in lieu of fractional shares of Qwest Common Stock, and unpaid dividends and distributions on shares of Qwest Common Stock deliverable in respect thereof, pursuant to this Agreement.

2.10 Withholding Rights. Each of the Surviving Corporation and Qwest shall be entitled to deduct and withhold from the consideration otherwise payable pursuant to this Agreement to any holder of shares of LCI Common Stock such amounts as it is required to deduct and withhold with respect to the making of such payment under the Code and the rules and regulations promulgated thereunder, or any provision of state, local or foreign tax law. To the extent that amounts are so withheld by the Surviving Corporation or Qwest, as the case may be, such withheld amounts shall be treated for all purposes of this Agreement as having been paid to the holder of the shares of LCI Common Stock in respect of which such deduction and withholding was made by the Surviving Corporation or Qwest, as the case may be.

2.11 Further Assurances. At and after the Effective Time, the officers and directors of the Surviving Corporation will be authorized to execute and deliver, in the name and on behalf of LCI or Merger Sub, any deeds, bills of sale, assignments or assurances and to take and do, in the name and on behalf of LCI or Merger Sub, any other actions and things to vest, perfect or confirm of record or otherwise in the Surviving Corporation any and all right, title and interest in, to and under any of the rights, properties or assets acquired or to be acquired by the Surviving Corporation as a result of, or in connection with, the Merger.

2.12 Stock Transfer Books. At the close of business, New York City time, on the day the Effective Time occurs, the stock transfer books of LCI shall be closed and there shall be no further registration of transfers of shares of LCI Common Stock thereafter on the records of LCI. From and after the Effective Time, the holders of Certificates shall cease to have any rights with respect to such shares of LCI Common Stock formerly represented thereby, except as otherwise provided herein or by law. On or after the Effective Time, any Certificates presented to the Exchange Agent or Qwest for any reason shall be converted into the Merger Consideration with respect to the shares of LCI Common Stock formerly represented thereby, any cash in lieu of fractional shares of Qwest Common Stock to which the holders thereof are entitled pursuant to Section 2.5 and any dividends or other distributions to which the holders thereof are entitled pursuant to Section 2.3.

ARTICLE III

REPRESENTATIONS AND WARRANTIES

3.1 Representations and Warranties of LCI. Except as set forth in the LCI Disclosure Schedule delivered by LCI to Qwest prior to the execution of this Agreement (the "LCI Disclosure Schedule") (each section of which qualifies the correspondingly numbered representation and warranty or covenant to the extent specified therein), LCI represents and warrants to Qwest as follows:

(a) Organization, Standing and Power. Each of LCI and each of its Subsidiaries is a corporation duly organized, validly existing and in good standing under the laws of its jurisdiction of incorporation or organization, has all requisite power and authority to own, lease and operate its properties and to carry on its business as now being conducted and is duly qualified and in good standing to do business in each jurisdiction in which the nature of its business or the ownership or leasing of its properties makes such qualification necessary other than in such jurisdictions where the failure so to qualify would not, either individually or in the aggregate, have a Material Adverse Effect (as defined in Section 8.11(d)) on LCI. The copies of the certificate of incorporation and by-laws of LCI which were previously furnished to Qwest are true, complete and correct copies of such documents as in effect on the date of this Agreement.

(b) Capital Structure.

As of February 28, 1998, the authorized capital (i) stock of LCI consisted of (A) 300,000,000 shares of LCI Common Stock, of which 96,814,777 shares were outstanding and (B) 15,000,000 shares of Preferred Stock, par value \$.01 per share, of which no shares were outstanding and 500,000 shares of which have been designated Junior Participating Preferred Stock and reserved for issuance upon exercise of the rights (the "Rights") distributed to the holders of LCI Common Stock pursuant to Rights Agreement dated as of January 22, 1997, between LCI and Fifth Third Bank, as Rights Agent, as amended (the "Rights Agreement"). Since February 28, 1998 to the date of this Agreement, there have been no issuances of shares of the capital stock of LCI or any other securities of LCI other than issuances of shares (and accompanying Rights) pursuant to options or rights outstanding as of February 28, 1998 under the Benefit Plans (as defined in Section 8.11(a)) of LCI. All issued and outstanding shares of the capital stock of LCI are duly authorized, validly

issued, fully paid and nonassessable, and no class of capital stock is entitled to preemptive rights. There were outstanding as of February 28, 1998 no options, warrants or other rights to acquire capital stock from LCI other than (x) the Rights, (y) options representing in the aggregate the right to purchase 14,302,838 shares

of LCI Common Stock (collectively, the "LCI Stock Options") under the LCI International, Inc. 1992 Stock Option Plan, LiTel Communications, Inc. 1993 Stock Option Plan, LCI International, Inc. 1994/1995 Stock Option Plan, LCI International, Inc. 1995/1996 Stock Option Plan, 1997/1998 LCI International, Inc. Stock Option Plan, LCI International, Inc. and Subsidiaries Nongualified Stock Option Plan for Directors, USLD Communications Corp. 1990 Employee Stock Option Plan, and USLD Communications Corp. 1993 Non-Employee Director Plan, (collectively, the "LCI Stock Option Plans"), and (z) a warrant representing the right to purchase 75,760 shares of LCI Common Stock (the "LCI Warrant"). As of February 28, 1998, LCI had further reserved 623,081 shares of LCI Common Stock for purchase pursuant to the LCI International, Inc. Amended and Restated Employee Stock Purchase Plan and 474,393 shares of LCI Common Stock for purchase pursuant to the LCI International 401(k) Savings Plan. Other than the associated Rights issued with the shares issued as described above and options or warrants or other rights to acquire no more than 100,000 shares of LCI Common Stock in the aggregate, no options or warrants or other rights to acquire capital stock from LCI have been issued or granted since February 28, 1998 to the date of this Agreement.

(ii) As of the date of this Agreement, no bonds, debentures, notes or other indebtedness of LCI having the right to vote on any matters on which stockholders may vote ("LCI Voting Debt") are issued or outstanding.

(iii) Except as otherwise set forth in this Section 3.1(b) and as contemplated by Section 5.6, as of the date of this Agreement, there are no securities, options, warrants, calls, rights, commitments, agreements, arrangements or undertakings of any kind to which LCI or any of its Subsidiaries is a party or by which any of them is bound obligating LCI or any of its Subsidiaries to issue, deliver or sell, or cause to be issued, delivered or sold, additional shares of capital stock or other voting securities of LCI or any of its Subsidiaries or obligating LCI or any of its Subsidiaries to issue, grant, extend or enter into any such security, option, warrant, call, right, commitment, agreement, arrangement or undertaking. As of the date of this Agreement, there are no outstanding obligations of LCI or any of its Subsidiaries to repurchase, redeem or otherwise acquire any shares of capital stock of LCI or any of its Subsidiaries.

(c) Authority; No Conflicts.

(i) LCI has all requisite corporate power and authority to enter into this Agreement and to consummate the transactions contemplated hereby, subject in the case of the consummation of the Merger to the adoption of this Agreement by the Required LCI Vote (as defined in Section 3.1(g)). The execution and delivery of this Agreement and the consummation of the transactions contemplated hereby have been duly authorized by all necessary corporate action on

the part of LCI, subject in the case of the consummation of the Merger to the adoption of this Agreement by the Required LCI Vote. This Agreement has been duly executed and delivered by LCI and constitutes a valid and binding agreement of LCI, enforceable against it in accordance with its terms, except as such enforceability may be limited by bankruptcy, insolvency, reorganization, moratorium and similar laws relating to or affecting creditors generally, by general equity principles (regardless of whether such enforceability is considered in a proceeding in equity or at law) or by an implied covenant of good faith and fair dealing.

(ii) The execution and delivery of this Agreement by LCI does not or will not, as the case may be, and the consummation of the Merger by LCI and the other transactions contemplated hereby will not, conflict with, or result in any violation of, or constitute a default (with or without notice or lapse of time, or both) under, or give rise to a right of termination, amendment, cancellation or acceleration of any obligation or the loss of a material benefit under, or the creation of a lien, pledge, security interest, charge or other encumbrance on any assets (any such conflict, violation, default, right of termination, amendment, cancellation or acceleration, loss or creation, a "Violation") pursuant to: (A) any provision of the certificate of incorporation or by-laws of LCI or any Subsidiary of LCI, or (B) except as would not have a Material Adverse Effect on LCI, subject to obtaining or making the consents, approvals, orders, authorizations, registrations, declarations and filings referred to in paragraph (iii) below, any loan or credit agreement, note, mortgage, bond, indenture, lease, benefit plan or other agreement, obligation, instrument, permit, concession, franchise, license, judgment, order, decree, statute, law, ordinance, rule or regulation applicable to LCI or any Subsidiary of LCI or their respective properties or assets.

(iii) No consent, approval, order or authorization of, or registration, declaration or filing with, any supranational, national, state, municipal or local government, any instrumentality, subdivision, court, administrative agency or commission or other authority thereof, or any quasi-governmental or private body exercising any regulatory, taxing, importing or other governmental or quasi-governmental authority (a "Governmental Entity"), is required by or with respect to LCI or any Subsidiary of LCI in connection with the execution and delivery of this Agreement by LCI or the consummation of the Merger and the other transactions contemplated hereby, except for those required under or in relation to (A) the Hart-Scott-Rodino Antitrust Improvements Act of 1976, as amended (the "HSR Act"), (B) the Communications Act of 1934, as amended (the "Communications Act"), and any rules, regulations, practices and policies promulgated by the Federal Communications Commission ("FCC"), (C) state securities or "blue sky" laws (the "Blue Sky Laws"), (D) the Securities Act of 1933, as amended (the "Securities Act"), (E) the Securities Exchange Act of 1934, as amended (the "Exchange Act"), (F) the DGCL with respect to the filing of the Delaware Certificate of Merger, (G) laws, rules, regulations, practices and orders of any state public service commissions ("PUCs"), foreign telecommunications regulatory agencies or similar state or foreign regulatory bodies, (H) rules and regulations of NASDAQ and the New York Stock Exchange, Inc. (the "NYSE"), (I)

antitrust or other competition laws of other jurisdictions, and (J) such consents, approvals, orders, authorizations, registrations, declarations and filings the failure of which to make or obtain would not have a Material Adverse Effect on LCI. Consents, approvals, orders, authorizations, registrations, declarations and filings required under or in relation to any of the foregoing clauses (A) through (I) are hereinafter referred to as "Required Consents."

(d) Reports and Financial Statements.

(i) LCI has filed all required reports, schedules, forms, statements and other documents required to be filed by it with the Securities and Exchange Commission (the "SEC") since January 1, 1997 (collectively, including all exhibits thereto, the "LCI SEC Reports"). No Subsidiary of LCI is required to file any form, report or other document with the SEC. None of the LCI SEC Reports, as of their respective dates (and, if amended or superseded by a filing prior to the date of this Agreement or the Closing Date, then on the date of such filing), contained or will contain any untrue statement of a material fact or omitted or will omit to state a material fact required to be stated therein or necessary to make the statements therein, in light of the circumstances under which they were made, not misleading. Each of the financial statements (including the related notes) included in the LCI SEC Reports presents fairly, in all material respects, the consolidated financial position and consolidated results of operations and cash flows of LCI and its Subsidiaries as of the respective dates or for the respective periods set forth therein, all in conformity with United States generally accepted accounting principles ("GAAP") consistently applied during the periods involved except as otherwise noted therein, and subject, in the case of the unaudited interim financial statements, to normal and recurring year-end adjustments that have not been and are not expected to be material in amount. All of such LCI SEC Reports, as of their respective dates (and as of the date of any amendment to the respective LCI SEC Report), complied as to form in all material respects with the applicable requirements of the Securities Act and the Exchange Act and the rules and regulations promulgated thereunder.

(ii) LCI has made available to Qwest drafts of the consolidated financial statements of LCI and its Subsidiaries at and

for the year ended December 31, 1997 of LCI and a proxy statement of LCI with respect to the proposed 1998 annual meeting of stockholders of LCI (each in the respective form thereof as of the date of this Agreement, collectively, the "LCI Draft Disclosures"). To the knowledge of LCI, each of the LCI Draft Disclosures is in substantially final form, except that the LCI Draft Disclosures do not disclose any information with respect to this Agreement, the transactions contemplated hereby or the effect that this Agreement or such transactions might have on the business, financial condition or results of operations (actual, pro forma or projected) of LCI and its Subsidiaries (collectively, the "LCI Transaction Information"). The LCI Draft Disclosures were not prepared for the purpose of providing to Qwest or any other Person any LCI Transaction Information. To the knowledge of LCI, except in each case with respect to LCI Transaction Information, (x) the draft proxy statement included in the LCI Draft Disclosures, as of the date of this Agreement, does not contain any

untrue statement of a material fact and does not omit to state a material fact required to be stated therein or necessary to make the statements therein, in light of the circumstances under which they are made, not misleading, (y) the draft financial statements (including the related notes) included in the LCI Draft Disclosures present fairly, in all material respects, the consolidated financial position and consolidated results of operations and cash flows of LCI and its Subsidiaries as of the respective dates or for the respective periods set forth therein, all in conformity with GAAP consistently applied during the periods involved except as otherwise noted therein and (z) all of the LCI Draft Disclosures comply as to form in all material respects with the applicable requirements of the Exchange Act and the rules and regulations promulgated thereunder.

(iii) From December 31, 1997 until the date of this Agreement, LCI and its Subsidiaries have not incurred any liabilities that are of a nature that would be required to be disclosed on a balance sheet of LCI and its Subsidiaries or the footnotes thereto prepared in conformity with GAAP, other than (A) liabilities incurred in the ordinary course of business or (B) liabilities that would not have a Material Adverse Effect on LCI and its Subsidiaries, taken together.

(e) Information Supplied.

(i) None of the information supplied or to be supplied by LCI for inclusion or incorporation by reference in (A) the registration statement on Form S-4 (as defined in Section 5.1) to be filed with the SEC by Qwest in connection with the issuance of the Qwest Common Stock in the Merger will, at the time the Form S-4 is filed with the SEC, at any time it is amended or supplemented or at the time it becomes effective under the Securities Act, contain any untrue statement of a material fact or omit to state any material fact required to be stated therein or necessary to make the statements therein not misleading and (B) the Joint Proxy Statement/Prospectus (as defined in Section 5.1) included in the Form S-4 related to the LCI Stockholders Meeting and the Qwest Stockholders Meeting (each, as defined in Section 5.1) and the Qwest Common Stock to be issued in the Merger will, on the date it is first mailed to LCI stockholders or Qwest Stockholders or at the time of the LCI Stockholders Meeting or the Qwest Stockholders Meeting, contain any untrue statement of a material fact or omit to state any material fact required to be stated therein or necessary in order to make the statements therein, in light of the circumstances under which they were made, not misleading.

(ii) Notwithstanding the foregoing provisions of this Section 3.1(e), no representation or warranty is made by LCI with respect to statements made or incorporated by reference in the Form S-4 or the Joint Proxy Statement/Prospectus based on information supplied by Qwest for inclusion or incorporation by reference therein.

(f) Board Approval. The Board of Directors of LCI, by resolutions duly adopted at a meeting duly called and held and not subsequently rescinded or modified in any way (the "Board Approval"), has duly (i) determined that this Agreement and the Merger are in the best

interests of LCI and its stockholders, (ii) approved this Agreement and the Merger and (iii) recommended that the stockholders of LCI approve this Agreement and the Merger. The LCI Board Approval constitutes approval of this Agreement and the Merger for purposes of Section 203 of the DGCL.

(g) Vote Required. The affirmative vote of the holders of a majority of the outstanding shares of LCI Common Stock to approve the Merger (the "Required LCI Vote") is the only vote of the holders of any class or series of LCI capital stock necessary to adopt this Agreement and approve the transactions contemplated hereby (assuming that Qwest is not an "interested stockholder" of LCI under Section 203 of the DGCL immediately before the execution and delivery of this Agreement).

(h) Rights Agreement. The Board of Directors of LCI has approved an amendment to the Rights Agreement to the effect that Qwest, Merger Sub and their affiliates shall not become an "Acquiring Person" (as defined in the Rights Agreement) by reason of the execution of the Merger Agreement or the consummation of the Merger.

(i) Brokers or Finders. No agent, broker, investment banker, financial advisor or other firm or Person is or will be entitled to any broker's or finder's fee or any other similar commission or fee in connection with any of the transactions contemplated by this Agreement, except Lehman Brothers Inc. (the "LCI Financial Advisor"), whose fees and expenses will be paid by LCI in accordance with LCI's agreement with such firm, based upon arrangements made by or on behalf of LCI and previously disclosed to Qwest. (j) Opinion of LCI Financial Advisor. LCI has received the opinion of the LCI Financial Advisor, dated the date of this Agreement, to the effect that, as of such date, the Exchange Ratio is fair, from a financial point of view, to the holders of LCI Common Stock, a copy of which opinion has been made available to Qwest.

(k) Affiliate Letter. On or prior to the date of the LCI Stockholder Meeting, LCI will deliver to Qwest a letter (the "LCI Affiliate Letter") identifying all persons who are "affiliates" of LCI for purposes of Rule 145 under the Securities Act ("Rule 145"). On or prior to the Closing Date, LCI will use all reasonable efforts to cause each person identified as an "affiliate" in the LCI Affiliate Letter to deliver a written agreement (an "Affiliate Agreement") in connection with restrictions on affiliates under Rule 145.

3.2 Representations and Warranties of Qwest. Qwest represents and warrants to LCI as follows:

(a) Organization, Standing and Power. Each of Qwest and each of its Subsidiaries is a corporation duly organized, validly existing and in good standing under the laws of its jurisdiction of incorporation or organization, has all requisite power and authority to own, lease and operate its properties and to carry on its business as now being conducted and is duly qualified and in good standing to do business in each jurisdiction in which the nature of its business or the ownership or leasing of its properties makes such qualification necessary other than in such jurisdictions where the failure so to qualify or to be in good standing would not, either individually or in the aggregate, have a Material Adverse Effect on Qwest. The copies of the certificate of incorporation and by-laws of

Qwest which were previously furnished to LCI are true, complete and correct copies of such documents as in effect on the date of this Agreement.

(b) Capital Structure.

As of February 28, 1998, the authorized capital (i) stock of Qwest consisted of (A) 400,000,000 shares of Qwest Common Stock of which 206,669,874 shares were outstanding and in excess of 80% of such outstanding shares were held of record by the Principal Stockholder as of such date and (B) 25,000,000 shares of Preferred Stock, par value \$.01 per share, of which no shares were outstanding. Since February 28, 1998 to the date of this Agreement, there have been no issuances of shares of the capital stock of Qwest or any other securities of Qwest other than issuances of shares pursuant to options or rights outstanding as of February 28, 1998 under the Benefit Plans (as defined in Section 8.11(h)) of Qwest. All issued and outstanding shares of the capital stock of Qwest are duly authorized, validly issued, fully paid and nonassessable, and no class of capital stock is entitled to preemptive rights. There were outstanding as of February 28, 1998 no options, warrants or other rights to acquire capital stock from Qwest other than (x) options

representing in the aggregate the right to purchase 15,374,000 shares of Qwest Common Stock issued to current or former employees and directors of Qwest and its Subsidiaries pursuant to Qwest's Equity Incentive Plan, (y) warrants representing in the aggregate the right to purchase 8,600,000 shares of Qwest Common Stock issued by Qwest to Anschutz Family Investment Company LLC (the "Qwest Warrants") and (z) rights to acquire shares of Qwest Common Stock pursuant to the Amended and Restated Agreement and Plan of Merger dated as of December 3, 1997 among Phoenix Network, Inc., Qwest and a subsidiary of Qwest. Other than options or warrants or other rights to acquire no more than 1,800,000 shares of Qwest Common Stock in the aggregate, no options or warrants or other rights to acquire capital stock from Qwest have been issued or granted since February 28, 1998 to the date of this Agreement.

(ii) As of the date of this Agreement, no bonds, debentures, notes or other indebtedness of Qwest having the right to vote on any matters on which stockholders may vote ("Qwest Voting Debt") are issued or outstanding.

(iii) Except as otherwise set forth in this Section 3.2(b), as of the date of this Agreement, there are no securities, options, warrants, calls, rights, commitments, agreements, arrangements or undertakings of any kind to which Qwest or any of its Subsidiaries is a party or by which any of them is bound obligating Qwest or any of its Subsidiaries to issue, deliver or sell, or cause to be issued, delivered or sold, additional shares of capital stock or other voting securities of Qwest or any of its Subsidiaries or obligating Qwest or any of its Subsidiaries to issue, grant, extend or enter into any such security, option, warrant, call, right, commitment, agreement, arrangement or undertaking. As of the date of this Agreement, there are no outstanding obligations of Qwest or any of its Subsidiaries to repurchase, redeem or otherwise acquire any shares of capital stock of Qwest or any of its Subsidiaries.

(c) Authority; No Conflicts.

(i) Qwest has all requisite corporate power and authority to enter into this Agreement and to consummate the transactions contemplated hereby, subject, in the case of the issuance of the shares of Qwest Common Stock to be issued in the Merger (the "Share Issuance"), to the adoption of this Agreement and the adoption of an amendment to the certificate of incorporation of Qwest increasing the number of authorized shares of Qwest Common Stock to not less than 600,000,000 shares (the "Qwest Charter Amendment") by the Required Qwest Vote (as defined in Section 3.2(g)) and the filing of the Qwest Charter Amendment with the Delaware Secretary of State. The execution and delivery of this Agreement and the consummation of the transactions contemplated hereby have been duly authorized by all necessary corporate action on the part of Qwest, subject to the approval by the stockholders of Qwest of the Share Issuance. This Agreement has been duly executed and delivered by Qwest and constitutes a valid and binding agreement of Qwest, enforceable against it in accordance with its terms, except as such enforceability may be limited by bankruptcy, insolvency, reorganization, moratorium and similar laws relating to or affecting creditors generally, by general equity principles (regardless of whether such enforceability is considered in a proceeding in equity or at law) or by an implied covenant of good faith and fair dealing.

(ii) The execution and delivery of this Agreement by Qwest does not or will not, as the case may be, and the consummation by Qwest of the Merger and the other transactions contemplated hereby will not, conflict with, or result in a Violation pursuant to: (A) any provision of the certificate of incorporation or by-laws of Qwest or any Subsidiary of Qwest, (B) except as would not have a Material Adverse Effect on Qwest, subject to obtaining or making the consents, approvals, orders, authorizations, registrations, declarations and filings referred to in paragraph (iii) below, any loan or credit agreement, note, mortgage, bond, indenture, lease, benefit plan or other agreement, obligation, instrument, permit, concession, franchise, license, judgment, order, decree, statute, law, ordinance, rule or regulation applicable to Qwest or any Subsidiary of Qwest or their respective properties or assets.

(iii) No consent, approval, order or authorization of, or registration, declaration or filing with, any Governmental Entity is required by or with respect to Qwest or any Subsidiary of Qwest in connection with the execution and delivery of this Agreement by Qwest or the consummation of the Merger and the other transactions contemplated hereby, except for the Required Consents and such consents, approvals, orders, authorizations, registrations, declarations and filings the failure of which to make or obtain would not have a Material Adverse Effect on Qwest.

(d) Reports and Financial Statements.

(i) Qwest has filed all required reports, schedules, forms, statements and other documents required to be filed by it with the SEC since January 1, 1997 (collectively, including all exhibits thereto, the "Qwest SEC Reports"). No Subsidiary of Qwest is required to file any form, report or other document with the SEC. None of the Qwest SEC Reports, as of their respective dates (and, if amended or superseded by a filing prior to the date of this Agreement

or the Closing Date, then on the date of such filing), contained or will contain any untrue statement of a material fact or omitted or will omit to state a material fact required to be stated therein or necessary to make the statements therein, in light of the circumstances under which they were made, not misleading. Each of the financial statements (including the related notes) included in the Qwest SEC Reports presents fairly, in all material respects, the consolidated financial position and consolidated results of operations and cash flows of Qwest and its Subsidiaries as of the respective dates or for the respective periods set forth therein, all in conformity with GAAP consistently applied during the periods involved except as otherwise noted therein, and subject, in the case of the unaudited interim financial statements, to normal and recurring year-end adjustments that have not been and are not expected to be material in amount. All of such Qwest SEC Reports, as of their respective dates (and as of the date of any amendment to the respects with the applicable requirements of the Securities Act and the Exchange Act and the rules and regulations promulgated thereunder.

Quest has made available to LCI drafts of the (ii) consolidated financial statements of Qwest and its Subsidiaries at and for the year ended December 31, 1997 of Qwest and a proxy statement of Qwest with respect to the proposed 1998 annual meeting of stockholders of Qwest (each in the respective form thereof as of the date of this Agreement, collectively, the "Qwest Draft Disclosures"). To the knowledge of Qwest, each of the Qwest Draft Disclosures, including the financial statements included therein, is in substantially final form, except that the Qwest Draft Disclosures do not disclose any information with respect to this Agreement, the transactions contemplated hereby or the effect that this Agreement or such transactions might have on the business, financial condition or results of operations (actual, pro forma or projected) of Qwest and its Subsidiaries (collectively, the "Qwest Transaction Information"). The Qwest Draft Disclosures were not prepared for the purpose of providing to LCI or any other Person any Qwest Transaction Information. To the knowledge of Qwest, except in each case with respect to Qwest Transaction Information, (x) the draft proxy statement included in the Qwest Draft Disclosures, as of the date of this Agreement, does not contain any untrue statement of a material fact and does not omit to state a material fact required to be stated therein or necessary to make the statements therein, in light of the circumstances under which they are made, not misleading, (y) the draft financial statements (including the related notes) included in the Qwest Draft Disclosures present fairly, in all material respects, the consolidated financial position and consolidated results of operations and cash flows of Qwest and its Subsidiaries as of the respective dates or for the respective periods set forth therein, all in conformity with GAAP consistently applied during the periods involved except as otherwise noted therein and (z) all of the Qwest Draft Disclosures comply as to form in all material respects with the applicable requirements of the Exchange Act and the rules and regulations promulgated thereunder.

(iii) From December 31, 1997 until the date of this Agreement, Qwest and its Subsidiaries have not incurred any

liabilities that are of a nature that would be required to be disclosed on a balance sheet of Qwest and its Subsidiaries or the footnotes thereto prepared in conformity with GAAP, other than (A) liabilities incurred in the ordinary course of business or (B) liabilities that would not have a Material Adverse Effect on Qwest and its Subsidiaries, taken together.

(e) Information Supplied.

None of the information supplied or to be supplied (i) by Qwest for inclusion or incorporation by reference in (A) the Form S-4 will, at the time the Form S-4 is filed with the SEC, at any time it is amended or supplemented or at the time it becomes effective under the Securities Act, contain any untrue statement of a material fact or omit to state any material fact required to be stated therein or necessary to make the statements therein not misleading, and (B) the Joint Proxy Statement/Prospectus will, on the date it is first mailed to LCI stockholders or Qwest stockholders or at the time of the LCI Stockholders Meeting or the Owest Stockholders Meeting, contain any untrue statement of a material fact or omit to state any material fact required to be stated therein or necessary in order to make the statements therein, in light of the circumstances under which they were made, not misleading. The Form S-4 and the Joint Proxy Statement/Prospectus will comply as to form in all material respects with the requirements of the Exchange Act and the Securities Act and the rules and regulations of the SEC thereunder.

(ii) Notwithstanding the foregoing provisions of this Section 3.2(e), no representation or warranty is made by Qwest with respect to statements made or incorporated by reference in the Form S-4 or the Joint Proxy Statement/Prospectus based on information supplied by LCI for inclusion or incorporation by reference therein.

(f) Board Approval. The Board of Directors of Qwest, by resolutions duly adopted at a meeting duly called and held and not subsequently rescinded or modified in any way, has duly (i) determined that this Agreement and the Merger are in the best interests of Qwest and its stockholders, (ii) approved this Agreement and the Merger, (iii) approved the Qwest Charter Amendment and (iv) recommended that the stockholders of Qwest approve the Qwest Charter Amendment and the Share Issuance.

(g) Vote Required. The affirmative vote of holders of shares of Qwest Common Stock representing a majority of the outstanding Qwest Common Stock (in the case of the Qwest Charter Amendment) and of the total votes cast at a meeting of the holders of outstanding shares of Qwest Common Stock (in the case of the Share Issuance) (collectively, the "Required Qwest Vote"), is the only vote of the holders of any class or series of Qwest capital stock necessary to approve the Qwest Charter Amendment and the Share Issuance.

(h) Brokers or Finders. No agent, broker, investment banker,

financial advisor or other firm or Person is or will be entitled to any broker's or finder's fee or any other similar commission or fee in connection with any of the transactions contemplated by this Agreement based upon arrangements made by or on behalf of Qwest, except Salomon Brothers Inc (the "Qwest Financial Advisor"), whose fees and expenses will be paid by Qwest in

accordance with Qwest's agreement with such firm based upon arrangements made by or on behalf of Qwest and previously disclosed to LCI.

(i) Opinion of Financial Advisor. Qwest has received the opinion of the Qwest Financial Advisor, dated the date of this Agreement, to the effect that, as of such date, the Exchange Ratio is fair, from a financial point of view, to Qwest, a copy of which opinion has been made available to LCI.

3.3 Representations and Warranties of Qwest and Merger Sub. Qwest and Merger Sub represent and warrant to LCI as follows:

(a) Organization and Corporate Power. Merger Sub is a corporation duly incorporated, validly existing and in good standing under the laws of Delaware. Merger Sub is a direct wholly-owned subsidiary of Qwest.

(b) Corporate Authorization. Merger Sub has all requisite corporate power and authority to enter into this Agreement and to consummate the transactions contemplated hereby. The execution, delivery and performance by Merger Sub of this Agreement and the consummation by Merger Sub of the transactions contemplated hereby have been duly authorized by all necessary corporate action on the part of Merger Sub. This Agreement has been duly executed and delivered by Merger Sub and constitutes a valid and binding agreement of Merger Sub, enforceable against it in accordance with its terms, except as such enforceability may be limited by bankruptcy, insolvency, reorganization, moratorium and other similar laws relating to or affecting creditors generally, by general equity principles (regardless or whether such enforceability is considered in a proceeding in equity or at law) or by an implied covenant of good faith and fair dealing.

(c) Non-Contravention. The execution, delivery and performance by Merger Sub of this Agreement and the consummation by Merger Sub of the transactions contemplated hereby do not and will not contravene or conflict with the certificate of incorporation or by-laws of Merger Sub.

(d) No Business Activities. Merger Sub has not conducted any activities other than in connection with the organization of Merger Sub, the negotiation and execution of this Agreement and the consummation of the transactions contemplated hereby. Merger Sub has no Subsidiaries.

ARTICLE IV

COVENANTS RELATING TO CONDUCT OF BUSINESS

4.1 Covenants of LCI. During the period from the date of this Agreement and continuing until the Effective Time, LCI agrees as to itself and its Subsidiaries that (except as expressly contemplated or permitted by this Agreement or as otherwise indicated on the LCI Disclosure Schedule or as required by a Governmental Entity of competent jurisdiction or to the extent that Qwest shall otherwise consent in writing):

(a) Ordinary Course.

(i) LCI and its Subsidiaries shall carry on their respective businesses in the usual, regular and ordinary course in

all material respects, in substantially the same manner as heretofore conducted, and shall use all reasonable efforts to preserve intact their present lines of business, maintain their rights and franchises and preserve their relationships with customers, suppliers and others having business dealings with them to the end that their ongoing businesses shall not be impaired in any material respect at the Effective Time; provided, however, that no action by LCI or its Subsidiaries with respect to matters specifically addressed by any other provision of this Section 4.1 shall be deemed a breach of this Section 4.1(a)(i) unless such action would constitute a breach of one or more of such other provisions.

(ii) LCI shall not, and shall not permit any of its Subsidiaries to, (A) enter into any new material line of business or (B) incur or commit to any capital expenditures other than capital expenditures incurred or committed to in the ordinary course of business consistent with past practice and which, together with all such expenditures incurred or committed since January 1, 1998, are not in excess of the respective amounts by category or in the aggregate set forth in LCI's 1998 capital expenditure budget, as previously disclosed to Qwest or, if the Closing Date has not occurred prior to December 31, 1998, such additional amounts for any subsequent period as may be consented to by Qwest, such consent not to be unreasonably withheld, or, if Qwest shall not have so consented, an amount not greater than an amount equal to a pro rata portion of LCI's 1998 Capital Expenditure Budget.

(b) Dividends; Changes in Share Capital. LCI shall not, and shall not permit any of its Subsidiaries to, and shall not propose to, (i) declare or pay any dividends on or make other distributions in respect of any of its capital stock, except dividends by wholly owned Subsidiaries of LCI, (ii) split, combine or reclassify any of its capital stock or issue or authorize or propose the issuance of any other securities in respect of, in lieu of or in substitution for, shares of its capital stock, except for any such transaction by a wholly owned Subsidiary of LCI which remains a wholly owned Subsidiary after consummation of such transaction, or (iii) repurchase, redeem or otherwise acquire any shares of its capital stock or any securities convertible into or exercisable for any shares of its capital stock except for the purchase from time to time by LCI of LCI Common Stock (and the associated Rights) in the ordinary course of business consistent with past practice in connection with the LCI Benefit Plans.

(C) Issuance of Securities. LCI shall not, and shall not permit any of its Subsidiaries to, issue, deliver or sell, or authorize or propose the issuance, delivery or sale of, any shares of its capital stock of any class, any LCI Voting Debt or any securities convertible into or exercisable for, or any rights, warrants or options to acquire, any such shares or LCI Voting Debt, or enter into any agreement with respect to any of the foregoing, other than (i) the issuance of LCI Common Stock (and the associated Rights) upon the exercise of LCI Warrants or stock options or in connection with other stock-based benefits plans outstanding on the date hereof in accordance with their present terms, (ii) issuances by a wholly owned Subsidiary of LCI of capital stock to such Subsidiary's parent, (iii) issuances in accordance with the Rights Agreement or (iv) issuances of shares, options, rights or other awards and amendments to equity-related awards in numbers not greater than those set forth in Section 4.1(c) of the LCI Disclosure Schedule.

(d) Governing Documents. Except to the extent required to comply with their respective obligations hereunder, required by law or required by the rules and regulations of the NYSE, LCI and its material Subsidiaries shall not amend, in the case of Subsidiaries, in any material respect, or propose to so amend their respective certificates of incorporation, by-laws or other governing documents.

No Acquisitions. LCI shall not, and shall not permit any (e) of its Subsidiaries to, acquire or agree to acquire by merging or consolidating with, or by purchasing a substantial equity interest in or a substantial portion of the assets of, or by any other manner, any business or any corporation, partnership, association or other business organization or division thereof or otherwise acquire or agree to acquire any assets (other than the acquisition of assets used in the operations of the business of LCI and its Subsidiaries in the ordinary course); provided, however, that the foregoing shall not prohibit (x) internal reorganizations or consolidations involving existing Subsidiaries of LCI, (y) the creation of new Subsidiaries of LCI organized to conduct or continue activities otherwise permitted by this Agreement or (z) after September 8, 1998, acquisitions by LCI and its Subsidiaries the fair market value of the total consideration (including the value of indebtedness or other liability acquired or assumed) for which does not exceed \$400 million in the aggregate.

(f) No Dispositions. Other than (i) internal reorganizations or consolidations involving existing Subsidiaries of LCI, (ii) dispositions referred to in LCI SEC Reports filed prior to the date of this Agreement, (iii) as may be required by or in conformance with law or regulation in order to permit or facilitate the consummation of the transactions contemplated hereby or (iv) in the ordinary course of business, LCI shall not, and shall not permit any Subsidiary of LCI to, sell, lease, encumber or otherwise dispose of, or agree to sell, lease, encumber or otherwise dispose of, any of its assets (including capital stock of Subsidiaries of LCI) which are material, individually or in the aggregate, to LCI.

Investments; Indebtedness. LCI shall not, and shall not (q) permit any of its Subsidiaries to, (i) other than in connection with actions permitted by Section 4.1(e), make any loans, advances or capital contributions to, or investments in, any other Person, other than by LCI or a Subsidiary of LCI to or in LCI or any Subsidiary of LCI, (ii) pay, discharge or satisfy any claims, liabilities or obligations (absolute, accrued, asserted or unasserted, contingent or otherwise), other than loans, advances, capital contributions, investments, payments, discharges or satisfactions incurred or committed to in the ordinary course of business consistent with past practice or (iii) other than in connection with actions permitted by Section 4.1(e), create, incur, assume or suffer to exist any indebtedness, issuances of debt securities, guarantees, loans or advances not in existence as of the date of this Agreement except pursuant to the credit facilities, indentures and other arrangements in existence on the date of this Agreement and in the ordinary course of business, and any other indebtedness existing on the date of this Agreement, in each case as such credit facilities, indentures, other arrangements and other existing indebtedness may be amended, extended, modified, refunded, renewed or refinanced after the date of this Agreement, but only if the aggregate principal amount thereof is not increased thereby, the term thereof is not extended thereby and the other terms and conditions thereof, taken as a whole, are not less advantageous to LCI and its Subsidiaries than those in existence as of the date of this Agreement.

(h) Tax-Free Qualification. LCI shall not, and shall not permit any of its Subsidiaries to, take any action that would prevent or impede the Merger from qualifying as a reorganization under Section 368 of the Code.

(i) Compensation. Other than as contemplated by Section 5.6 or by Schedule 4.1(i), LCI shall not, and shall not permit any of its Subsidiaries to, increase the amount of compensation of any senior executive officer except in the ordinary course of business consistent with past practice or as required by an existing agreement, make any increase in or commitment to increase any employee benefits, issue any additional LCI Stock Options, adopt or make any commitment to adopt any additional employee benefit plan or make any contribution, other than regularly scheduled contributions, to any Benefit Plan.

(j) Other Actions. LCI shall not, and shall not permit any of its Subsidiaries to, take any action that would, or that could reasonably be expected to, result in, except as otherwise permitted by Section 5.5, any of the conditions to the Merger set forth in Article VI not being satisfied.

(k) Accounting Methods; Income Tax Elections. Except as disclosed in LCI SEC Reports filed prior to the date of this Agreement, or as required by a Governmental Entity, LCI shall not change its methods of accounting in effect at December 31, 1997, except as required by changes in GAAP as concurred in by LCI's independent auditors. LCI shall not (i) change its fiscal year or (ii) make any material tax election, other than in the ordinary course of business consistent with past practice, without consultation with Qwest.

(1) Rights Agreement. Except as provided in Section 5.9, LCI shall not amend, modify or waive any provision of the Rights Agreement, and shall not take any action to redeem the Rights or render the Rights inapplicable to any transaction, other than to permit another transaction that the LCI Board has determined is a Superior Proposal (as defined in Section 8.11), to be consummated after termination of this Agreement.

4.2 Covenants of Qwest. During the period from the date of this Agreement and continuing until the Effective Time, Qwest agrees as to itself and its Subsidiaries that (except as expressly contemplated or permitted by this Agreement or as otherwise indicated on the Qwest Disclosure Schedule or as required by a Governmental Entity of competent jurisdiction or to the extent that LCI shall otherwise consent in writing):

(a) Ordinary Course.

(i) Qwest and its Subsidiaries shall carry on their respective businesses in the usual, regular and ordinary course in all material respects, in substantially the same manner as heretofore conducted, and shall use all reasonable efforts to preserve intact their present lines of business, maintain their rights and franchises and preserve their relationships with customers, suppliers and others having business dealings with them to the end that their ongoing businesses shall not be impaired in any material respect at the Effective Time; provided, however, that no action by Qwest or its Subsidiaries with respect to matters specifically addressed by any other provisions of this Section 4.2 shall be deemed a breach of this

Section 4.2(a)(i) unless such action would constitute a breach of one or more of such other provisions.

(ii) Qwest shall not, and shall not permit any of its Subsidiaries to, enter into any new material line of business that is not part of the communications business, other than incidentally as part of a larger acquisition within an existing line of business.

(b) Dividends; Changes in Share Capital. Qwest shall not, and shall not permit any of its Subsidiaries to, and shall not propose to, (i) declare or pay any dividends on or make other distributions in respect of any of its capital stock, except dividends by wholly owned Subsidiaries of Qwest, (ii) split, combine or reclassify any of its capital stock or issue or authorize or propose the issuance of any other securities in respect of, in lieu of or in substitution for, shares of its capital stock, except for any such transaction by a wholly owned Subsidiary of Qwest which remains a wholly owned Subsidiary after consummation of such transaction, (iii) repurchase, redeem or otherwise acquire any shares of its capital stock or any securities convertible into or exercisable for any shares of its capital stock except for the purchase from time to time by Qwest of Qwest Common Stock in the ordinary course of business required by its 401(k) plan on a nondiscretionary basis or in connection with rights to purchase shares directly from holders who received such shares in an acquisition by Qwest previously disclosed to LCI and permitted by Section 4.2(e).

Issuance of Securities. Qwest shall not, and shall not (C) permit any of its Subsidiaries to, issue, deliver or sell, or authorize or propose the issuance, delivery or sale of, any shares of its capital stock of any class, any Qwest Voting Debt or any securities convertible into or exercisable for, or any rights, warrants or options to acquire, any such shares or Qwest Voting Debt, or enter into any agreement with respect to any of the foregoing, or amend the Qwest Warrants other than (i) the issuance of Qwest Common Stock upon the exercise of stock options, (ii) issuances by a wholly owned Subsidiary of Qwest of capital stock to such Subsidiary's parent or another wholly owned Subsidiary of Qwest, (iii) issuances of options, awards, and amendments to equity-related awards pursuant to Qwest benefit plans as in effect from time to time, or (iv) issuances in respect of any acquisitions, mergers, share exchanges, consolidations, business combinations or similar transactions by Qwest or its Subsidiaries permitted by Sections 4.2(e) and 4.2(k).

(d) Governing Documents. Except to the extent required to comply with their respective obligations hereunder, required by law or required by the rules and regulations of NASDAQ, Qwest and its material Subsidiaries shall not amend, in the case of Subsidiaries, in any material respect, or propose to so amend their respective certificates of incorporation, by-laws or other governing documents, except that Merger Sub may amend its certificate of incorporation to increase the number of authorized shares of its common stock.

(e) No Acquisitions. Other than acquisitions the fair market value of the total consideration (including the value of indebtedness or other liability acquired or assumed) for which does not exceed \$1 billion in the aggregate, Qwest shall not, and shall not permit any of its Subsidiaries to, acquire or agree to acquire by merging or consolidating with, or by purchasing a substantial equity interest in or a substantial portion of the assets of, or by any other manner, any business or any corporation,

partnership, association or other business organization or division thereof or otherwise acquire or agree to acquire any assets (other than the acquisition of assets used in the ordinary course); provided, however, that the foregoing shall not prohibit (x) internal reorganizations or consolidations involving existing Subsidiaries of Qwest or (y) the creation of new Subsidiaries of Qwest organized to conduct or continue activities otherwise permitted by this Agreement.

(f) No Dispositions. Other than (i) internal reorganizations or consolidations involving existing Subsidiaries of Qwest, (ii) dispositions referred to in Qwest SEC Reports filed prior to the date of this Agreement, (iii) as may be required by or in conformance with law or regulation in order to permit or facilitate the consummation of the transactions contemplated hereby or (iv) in the ordinary course of business, Qwest shall not, and shall not permit any Subsidiary of Qwest to, sell, lease, encumber or otherwise dispose of, or agree to sell, lease, encumber or otherwise dispose of, any of its assets (including capital stock of Subsidiaries of Qwest) which are material, individually or in the aggregate, to Qwest.

(g) Investments. Other than in the ordinary course of business or as permitted by Section 4.2(e), Qwest shall not, and shall not permit any of its Subsidiaries to make any loans, advances or capital contributions to, or investments in, any other Person, other than by Qwest or a Subsidiary of Qwest to or in Qwest or any Subsidiary of Qwest.

(h) Tax-Free Qualification. Qwest shall not, and shall not permit any of its Subsidiaries to, take any action that would prevent or impede the Merger from qualifying as a reorganization under Section 368 of the Code.

(i) Other Actions. Qwest shall not, and shall not permit any of its Subsidiaries to, take any action that would, or that could reasonably be expected to, result in any of the conditions to the Merger set forth in Article VI not being satisfied.

(j) Accounting Methods; Income Tax Elections. Except as disclosed in Qwest SEC Reports filed prior to the date of this Agreement, or as required by a Governmental Entity, Qwest shall not change its methods of accounting in effect at December 31, 1997, except as required by changes in GAAP as concurred in by Qwest's independent auditors. Qwest shall not (i) change its fiscal year or (ii) make any material tax election, other than in the ordinary course of business consistent with past practice, without consultation with LCI.

(k) Qwest agrees that prior to the Closing Date it shall not, without the prior written consent of LCI (such consent not to be unreasonably withheld), agree to enter into any merger, reorganization, share exchange, business combination or similar transaction pursuant to which any stockholder of Qwest shall receive any consideration (whether payable in cash, securities, property or other consideration) in exchange for its shares of Qwest Common Stock unless both (i) such transaction is not to be consummated until after the termination of this Agreement pursuant to Section 7.1 or the Effective Time and (ii) the consideration per share of Qwest Common Stock payable in connection therewith has a value, as reasonably determined by Qwest, of not less than \$35-15/16.

4.3 Advice of Changes; Governmental Filings. Each party shall (a) confer on a regular and frequent basis with the other and (b) report (to the extent permitted by law or regulation or any applicable confidentiality agreement) on operational matters. LCI and Qwest shall file all reports required to be filed by each of them with the SEC (and all other Governmental Entities) between the date of this Agreement and the Effective Time and shall

(to the extent permitted by law or regulation or any applicable confidentiality agreement) deliver to the other party copies of all such reports, announcements and publications promptly after the same are filed. Subject to applicable laws relating to the exchange of information, each of LCI and Qwest shall have the right to review in advance, and will consult with the other with respect to, all the information relating to the other party and each of their respective Subsidiaries, which appears in any filings, announcements or publications made with, or written materials submitted to, any third party or any Governmental Entity in connection with the transactions contemplated by this Agreement. In exercising the foregoing right, each of the parties hereto agrees to act reasonably and as promptly as practicable. Each party agrees that, to the extent practicable and as timely as practicable, it will consult with, and provide all appropriate and necessary assistance to, the other party with respect to the obtaining of all permits, consents, approvals and authorizations of all third parties and Governmental Entities necessary or advisable to consummate the transactions contemplated by this Agreement and each party will keep the other party apprised of the status of matters relating to completion of the transactions contemplated hereby.

4.4 Transition Planning; Continued Operations of LCI. LCI and Qwest shall each appoint four officers to serve from time to time as their respective representatives on a committee that will be responsible for coordinating transition planning and implementation relating to the Merger. The initial representatives of LCI shall be H. Brian Thompson, Joseph Lawrence, Larry Bowman and Marsh Hanno. The initial representatives of Qwest shall be Joseph P. Nacchio, Robert S. Woodruff, Lewis O. Wilks and Larry A. Seese. It is the current intention of Qwest that LCI will continue to operate LCI's offices and facilities in Fairfax County, Virginia and Dublin, Ohio.

4.5 Control of Other Party's Business. Nothing contained in this Agreement shall give LCI, directly or indirectly, the right to control or direct Qwest's operations prior to the Effective Time. Nothing contained in this Agreement shall give Qwest, directly or indirectly, the right to control or direct LCI's operations prior to the Effective Time. Prior to the Effective Time, each of LCI and Qwest shall exercise, consistent with the terms and conditions of this Agreement, complete control and supervision over its respective operations.

ARTICLE V

ADDITIONAL AGREEMENTS

5.1 Preparation of Proxy Statement; LCI Stockholders Meeting.

(a) As promptly as practicable following the date hereof, Qwest shall, in cooperation with LCI, prepare and file with the SEC preliminary proxy materials which shall constitute the Joint Proxy Statement/Prospectus (such proxy statement/prospectus, and any amendments or

supplements thereto, the "Joint Proxy Statement/Prospectus") and a registration statement on Form S-4 with respect to the issuance of Qwest Common Stock in the Merger (the "Form S-4"). The Joint Proxy Statement/Prospectus will be included in the Form S-4 as Qwest's prospectus. The Form S-4 and the Joint Proxy Statement/Prospectus shall comply as to form in all material respects with the applicable provisions of the Securities Act and the Exchange Act and the rules and regulations thereunder. Each of Qwest and LCI shall use all reasonable efforts to have the Form S-4 cleared by the SEC as promptly as practicable after filing with the SEC and to keep the Form S-4 effective as long as is necessary to consummate the Merger. Qwest shall, as promptly as practicable after receipt thereof, provide copies of any written comments received from the SEC with respect to the Joint Proxy Statement/Prospectus to LCI and advise LCI of any oral comments with respect to the Proxy Statement/Prospectus received from the SEC. Qwest agrees that none of the information supplied or to be supplied by Qwest for inclusion or incorporation by reference in the Joint Proxy Statement/Prospectus and each amendment or supplement thereto, at the time of mailing thereof and at the time of the LCI Stockholders Meeting or the Qwest Stockholders Meeting, will contain an untrue statement of a material fact or omit to state a material fact required to be stated therein or necessary to make the statements therein, in light of the circumstances under which they were made, not misleading. LCI agrees that none of the information supplied or to be supplied by LCI for inclusion or incorporation by reference in the Joint Proxy Statement/Prospectus and each amendment or supplement thereto, at the time of mailing thereof and at the time of the LCI Stockholders Meeting or the Qwest Stockholders Meeting, will contain an untrue statement of a material fact or omit to state a material fact required to be stated therein or necessary to make the statements therein, in light of the circumstances under which they were made, not misleading. For purposes of the foregoing, it is understood and agreed that information concerning or related to Qwest and the Qwest Stockholders Meeting will be deemed to have been supplied by Qwest and information concerning or related to LCI and the LCI Stockholders Meeting shall be deemed to have been supplied by LCI. Qwest will provide LCI with a reasonable opportunity to review and comment on any amendment or supplement to the Joint Proxy Statement/Prospectus prior to filing such with the SEC, and will provide LCI with a copy of all such filings made with the SEC. No amendment or supplement to the information supplied by LCI for inclusion in the Joint Proxy Statement/Prospectus shall be made without the approval of LCI, which approval shall not be unreasonably withheld or delayed.

(b) Subject to Sections 5.5 and 7.1(f), LCI shall, as promptly as practicable following the execution of this Agreement, duly call, give notice of, convene and hold a meeting of its stockholders (the "LCI Stockholders Meeting") for the purpose of obtaining the Required LCI Vote with respect to the transactions contemplated by this Agreement, shall take all lawful action to solicit the adoption of this Agreement by the Required LCI Vote and the Board of Directors of LCI shall recommend adoption of this Agreement by the stockholders of LCI. Without limiting the generality of the foregoing but subject to its rights pursuant to Sections 5.5 and 7.1(f), LCI agrees that its obligations pursuant to the first sentence of this Section 5.1(b) shall not be affected by the commencement, public proposal, public disclosure or communication to LCI of any Acquisition Proposal.

(c) Qwest shall, as promptly as practicable following the execution of this Agreement, duly call, give notice of, convene and hold a meeting of its stockholders (the "Qwest Stockholders Meeting") for the

purpose of obtaining the Required Qwest Vote, shall take all lawful action to solicit the approval of the Share Issuance by the Required Qwest Vote and the Board of Directors of Qwest shall recommend approval of the transactions contemplated by this Agreement by the stockholders of Qwest.

5.2 Qwest Board of Directors. At or prior to the Effective Time, the Board of Directors of Qwest will take all action necessary to elect the Chief Executive Officer and one other director of LCI on the date of this Agreement selected by LCI as members of the Board of Directors of Qwest to serve until the end of the term beginning at the annual meeting of Qwest's stockholders in 1999. In the event that such Chief Executive Officer is so elected and agrees to serve as a director of Qwest, the Board of Directors of Qwest shall appoint him as Vice Chairman of Qwest.

Access to Information. Upon reasonable notice, each party 5.3 shall (and shall cause its Subsidiaries to) afford to the officers, employees, accountants, counsel, financial advisors and other representatives of the other party reasonable access during normal business hours, during the period prior to the Effective Time, to all its properties, books, contracts, commitments and records and, during such period, such party shall (and shall cause its Subsidiaries to) furnish promptly to the other party (a) a copy of each report, schedule, registration statement and other document filed, published, announced or received by it during such period pursuant to the requirements of Federal or state securities laws, as applicable (other than documents which such party is not permitted to disclose under applicable law), and (b) consistent with its legal obligations, all other information concerning its business, properties and personnel as such other party may reasonably request; provided, however, that either party may restrict the foregoing access to the extent that (i) a Governmental Entity requires such party or any of its Subsidiaries to restrict access to any properties or information reasonably related to any such contract on the basis of applicable laws and regulations with respect to national security matters or (ii) any law, treaty, rule or regulation of any Governmental Entity applicable to such party requires such party or its Subsidiaries to restrict access to any properties or information. The parties will hold any such information which is non-public in confidence to the extent required by, and in accordance with, the provisions of the letter dated February 25, 1998 between LCI and Qwest (the "Confidentiality Agreement"). Any investigation by Qwest or LCI shall not affect the representations and warranties of LCI or Qwest, as the case may be.

5.4 Best Efforts.

(a) Subject to the terms and conditions of this Agreement,

each party will use its best efforts to take, or cause to be taken, all actions and to do, or cause to be done, all things necessary, proper or advisable under applicable laws and regulations to consummate the Merger and the other transactions contemplated by this Agreement as soon as practicable after the date hereof. In furtherance and not in limitation of the foregoing, each party hereto agrees to make an appropriate filing of a Notification and Report Form pursuant to the HSR Act with respect to the transactions contemplated hereby as promptly as practicable after the date hereof and to supply as promptly as practicable any additional information and documentary material that may be requested pursuant to the HSR Act and to take all other actions necessary to cause the expiration or termination of the applicable waiting periods under the HSR Act as soon as practicable. Nothing in this Section 5.4(a) shall require any of Qwest and its

Subsidiaries to sell or otherwise dispose of, or permit the sale or other disposition of, any assets of Qwest, LCI or their respective Subsidiaries, whether as a condition to obtaining any approval from a Governmental Entity or any other Person or for any other reason, if Qwest reasonably determines that such sale or other disposition would have or is likely to have a Material Adverse Effect on Qwest and its Subsidiaries (including the Surviving Corporation and its Subsidiaries), taken together, after giving effect to the Merger.

Each of Qwest and LCI shall, in connection with the (b) efforts referenced in Section 5.4(a) to obtain all requisite approvals and authorizations for the transactions contemplated by this Merger Agreement under the HSR Act or any other Regulatory Law (as defined below), use its best efforts to (i) cooperate in all respects with each other in connection with any filing or submission and in connection with any investigation or other inquiry, including any proceeding initiated by a private party, (ii) promptly inform the other party of any communication received by such party from, or given by such party to, the FCC, PUCs, the Antitrust Division of the Department of Justice (the "DOJ") or any other Governmental Entity and of any material communication received or given in connection with any proceeding by a private party, in each case regarding any of the transactions contemplated hereby, and (iii) permit the other party to review any communication given by it to, and consult with each other in advance of any meeting or conference with, the FCC, PUCs, the DOJ or any such other Governmental Entity or, in connection with any proceeding by a private party, with any other Person, and to the extent permitted by the FCC, PUCs, the DOJ or such other applicable Governmental Entity or other Person, give the other party the opportunity to attend and participate in such meetings and conferences. For purposes of this Agreement, "Regulatory Law" means the Sherman Act, as amended, the Clayton Act, as amended, the HSR Act, the Federal Trade Commission Act, as amended, the Federal Communications Act, as amended, and all other federal, state and foreign, if any, statutes, rules, regulations, orders, decrees, administrative and judicial doctrines and other laws that are designed or intended to prohibit, restrict or regulate actions having the purpose or effect of monopolization or restraint of trade or lessening of competition, whether in the communications industry or otherwise through merger or acquisition.

In furtherance and not in limitation of the covenants of (C)the parties contained in Sections 5.4(a) and 5.4(b), if any administrative or judicial action or proceeding, including any proceeding by a private party, is instituted (or threatened to be instituted) challenging any transaction contemplated by this Agreement as violative of any Regulatory Law, each of Qwest and LCI shall cooperate in all respects with each other and use its respective best efforts to contest and resist any such action or proceeding and to have vacated, lifted, reversed or overturned any decree, judgment, injunction or other order, whether temporary, preliminary or permanent, that is in effect and that prohibits, prevents or restricts consummation of the transactions contemplated by this Agreement. Notwithstanding the foregoing or any other provision of this Agreement, nothing in this Section 5.4 shall limit a party's right to terminate this Agreement pursuant to Section 7.1(b) or 7.1(c) so long as such party has up to then complied in all respects with its obligations under this Section 5.4.

(d) If any objections are asserted with respect to the transactions contemplated hereby under any Regulatory Law or if any suit is instituted by any Governmental Entity or any private party challenging any of

the transactions contemplated hereby as violative of any Regulatory Law, each of Qwest and LCI shall use its best efforts to resolve any such objections or challenge as such Governmental Entity or private party may have to such transactions under such Regulatory Law so as to permit consummation of the transactions contemplated by this Agreement.

(e) Each of Qwest, Merger Sub and LCI shall use its best efforts to cause the Merger to qualify and will not (both before and after consummation of the Merger) take any actions which to its knowledge could reasonably be expected to prevent the Merger from qualifying, as a reorganization under the provisions of Section 368 of the Code.

5.5 Acquisition Proposals. LCI agrees that neither it nor any of its Subsidiaries nor any of the officers and directors of it or its Subsidiaries shall, and that it shall direct and use its best efforts to cause its and its Subsidiaries' employees, agents and representatives (including any investment banker, attorney or accountant retained by it or any of its Subsidiaries) not to, directly or indirectly, initiate, solicit, encourage or knowingly facilitate (including by way of furnishing information) any inquiries or the making of any proposal or offer with respect to a merger, reorganization, share exchange, consolidation, business combination, recapitalization, liquidation, dissolution or similar transaction involving, or any purchase or sale of all or any significant portion of the assets or more than 15% (or in the case of acquisition by an Institutional Investor (as defined in the Rights Plan) more than 20%) of the common stock of, it or any of its Subsidiaries (any such proposal or offer (other than a proposal or offer made by Qwest or an affiliate thereof) being hereinafter referred to as an "Acquisition Proposal"). LCI further agrees that neither it nor any of its Subsidiaries nor any of the officers and directors of it or its Subsidiaries shall, and that it shall direct and use

its best efforts to cause its and its Subsidiaries' employees, agents and representatives (including any investment banker, attorney or accountant retained by it or any of its Subsidiaries) not to, directly or indirectly, have any discussion with or provide any confidential information or data to any Person relating to an Acquisition Proposal, or engage in any negotiations concerning an Acquisition Proposal, or knowingly facilitate any effort or attempt to make or implement an Acquisition Proposal or accept an Acquisition Proposal. Notwithstanding the foregoing, LCI or its Board of Directors shall be permitted to (A) to the extent applicable, comply with Rule 14d-9 and Rule 14e-2(a) promulgated under the Exchange Act with regard to an Acquisition Proposal, (B) in response to an unsolicited bona fide written Acquisition Proposal by any Person, recommend approval of such an unsolicited bona fide written Acquisition Proposal to the stockholders of LCI or withdraw or modify in any adverse manner the Board Approval or (C) engage in any discussions or negotiations with, or provide any information to, any Person in response to an unsolicited bona fide written Acquisition Proposal by any such Person, if and only to the extent that, in any such case as is referred to in clause (B) or (C), (i) the LCI Stockholders Meeting shall not have occurred, (ii) the Board of Directors of LCI concludes in good faith that such Acquisition Proposal (x) in the case of clause (B) above would, if consummated, constitute a Superior Proposal or (y) in the case of clause (C) above could reasonably be expected to constitute a Superior Proposal, (iii) prior to providing any information or data to any Person in connection with an Acquisition Proposal by any such Person, the LCI Board of Directors receives from such Person an executed confidentiality agreement on terms substantially similar to those contained in the Confidentiality Agreement (except as to the standstill provisions, provided that if under the aforementioned

circumstances LCI enters into any such confidentiality agreement without standstill provisions substantially similar to those contained in the Confidentiality Agreement, then Owest shall to the extent of the difference be relieved of compliance with the Confidentiality Agreement's standstill provisions), and (iv) prior to providing any information or data to any Person or entering into discussions or negotiations with any Person, the Board of Directors of LCI notifies Qwest promptly of such inquiries, proposals or offers received by, any such information requested from, or any such discussions or negotiations sought to be initiated or continued with, any of its representatives indicating, in connection with such notice, the name of such Person and the material terms and conditions of any proposals or offers. LCI agrees that it will keep Qwest informed, on a current basis, of the status and terms of any such proposals or offers and the status of any such discussions or negotiations. LCI agrees that it will immediately cease and cause to be terminated any existing activities, discussions or negotiations with any parties conducted heretofore with respect to any Acquisition Proposal. LCI agrees that it will take the necessary steps to promptly inform the individuals or entities referred to in the first sentence of this Section 5.5 of the obligations undertaken in this Section 5.5. Nothing in this Section 5.5 shall (x) permit LCI to terminate this Agreement (except as specifically provided in Article VII hereof) or (y) affect any other obligation of LCI under this Agreement.

5.6 Assumption of LCI Stock Options; Other Stock Plans; Employee Benefits Matters.

(a) Qwest shall assume the LCI Stock Options on the terms set forth in Exhibit 5.6 hereto. LCI and Qwest agree that LCI's other stock plans and treatment of LCI's officers and employees shall be as set forth in Exhibit 5.6 hereto.

(b) Employee Benefits.

Obligations of Qwest; Comparability of Benefits. Each (i) Benefit Plan (as defined in Section 8.11(a)) as to which LCI or any of its Subsidiaries has any obligation with respect to any current or former employee (the "LCI Employees") (the "LCI Benefit Plans") shall be the obligations of Qwest and the Surviving Corporation at the Effective Time and for at least two years thereafter, Qwest shall, or shall cause the Surviving Corporation to, provide benefits, in the aggregate, that are no less favorable than the benefits provided, in the aggregate, under such Benefit Plans to the LCI Employees immediately prior to the Effective Time. Notwithstanding the foregoing, nothing herein shall require (A) the continuation of any particular LCI Benefit Plan or prevent the amendment or termination thereof (subject to the maintenance, in the aggregate, of the benefits as provided in the preceding sentence) or (B) require Qwest or the Surviving Corporation to continue or maintain any stock purchase or other equity plan related to the equity of LCI or the Surviving Corporation.

(ii) Pre-Existing Limitations; Deductible; Service Credit. With respect to any Benefit Plans of Qwest in which the LCI Employees participate effective as of the Closing Date, Qwest shall, or shall cause the Surviving Corporation to: (A) not impose any limitations more onerous than those currently in effect as to pre-existing conditions, exclusions and waiting periods with respect to

participation and coverage requirements applicable to the LCI Employees under which any welfare Benefit Plan in which such employees may be eligible to participate after the Effective Time, (B) provide each LCI Employee with credit for any co-payments and deductibles paid prior to the Effective Time in satisfying any applicable deductible or out-of-pocket requirements under any welfare Benefit Plan in which such employees may be eligible to participate after the Effective Time, and (C) recognize all service of the LCI Employees with LCI for all purposes (including, without limitation, purposes of eligibility to participate, vesting credit, entitlement for benefits, and benefit accrual) in any Benefit Plan in which such employees may be eligible to participate after the Effective Time, to the same extent taken into account under a comparable LCI Benefit Plan immediately prior to the Closing Date.

5.7 Fees and Expenses. Whether or not the Merger is

consummated, all Expenses incurred in connection with this Agreement and the transactions contemplated hereby shall be paid by the party incurring such Expenses, except (a) if the Merger is consummated, the Surviving Corporation shall pay, or cause to be paid, any and all property or transfer taxes imposed on LCI or its Subsidiaries and any real property transfer tax imposed on any holder of shares of capital stock of LCI resulting from the Merger, (b) Expenses incurred in connection with the filing, printing and mailing of the Joint Proxy Statement/Prospectus, which shall be shared equally by Qwest and LCI and (c) as provided in Section 7.2. As used in this Agreement, "Expenses" includes all out-of-pocket expenses (including, without limitation, all fees and expenses of counsel, accountants, investment bankers, experts and consultants to a party hereto and its affiliates) incurred by a party or on its behalf in connection with or related to the authorization, preparation, negotiation, execution and performance of this Agreement and the transactions contemplated hereby, including the preparation, printing, filing and mailing of the Joint Proxy Statement/Prospectus and the solicitation of stockholder approvals and all other matters related to the transactions contemplated hereby.

5.8 Directors' and Officers' Indemnification and Insurance. The Surviving Corporation shall cause to be maintained in effect in its certificate of incorporation and by-laws (i) for a period of six years after the Effective Time, the current provisions regarding elimination of liability of directors and indemnification of officers, directors and employees contained in the certificate of incorporation and by-laws of LCI and (ii) for a period of six years, the current policies of directors' and officers' liability insurance and fiduciary liability insurance maintained by LCI (provided that the Surviving Corporation may substitute therefor policies of at least the same coverage and amounts containing terms and conditions which are, in the aggregate, no less advantageous to the insured) with respect to claims arising from facts or events that occurred on or before the Effective Time; provided, however, that in no event shall the Surviving Corporation be required to expend in any one year an amount in excess of 200% of the annual premiums currently paid by LCI for such insurance; and, provided, further, that if the annual premiums of such insurance coverage exceed such amount, the Surviving Corporation shall be obligated to obtain a policy with the greatest coverage available for a cost not exceeding such amount.

5.9 Rights Agreement. The Board of Directors of LCI shall take all further action (in addition to that referred to in Section 3.1(h)) necessary (including redeeming the Rights immediately prior to the Effective

Time or amending the Rights Agreement) in order to render the Rights inapplicable to the Merger and the other transactions contemplated by this Agreement.

5.10 Public Announcements. LCI and Qwest shall use all reasonable efforts to develop a joint communications plan and each party shall use all reasonable efforts (i) to ensure that all press releases and other public statements with respect to the transactions contemplated hereby shall be consistent with such joint communications plan, and (ii) unless otherwise required by applicable law or by obligations pursuant to any listing agreement with or rules of any securities exchange, to consult with each other before issuing any press release or otherwise making any public statement with respect to this Agreement or the transactions contemplated hereby.

5.11 Accountants' Letters. Upon reasonable notice from the other, LCI and Qwest shall use their respective reasonable best efforts to cause Arthur Andersen LLP and KPMG Peat Marwick LLP, respectively, to deliver to LCI or Qwest, as the case may be, a letter, dated within two business days of the Effective Time of the Form S-4 covering such matters as are requested by Qwest or LCI, as the case may be, and as are customarily addressed in accountant's "comfort" letters. In connection with LCI's efforts to obtain such letter, if requested by Arthur Andersen LLP, Qwest shall provide a representation letter to Arthur Andersen LLP complying with the statement on Auditing Standards No. 72 ("SAS 72"), if then required. In connection with Qwest's efforts to obtain such letter, if requested by KPMG Peat Marwick LLP, LCI shall provide a representation letter to KPMG Peat Marwick LLP complying with SAS 72, if then required.

5.12 Listing of Shares of Qwest Common Stock. Qwest shall use its best efforts to cause the shares of Qwest Common Stock to be issued in the Merger and the shares of Qwest Common Stock to be reserved for issuance upon exercise of the LCI Stock Options and LCI Warrants to be approved for quotation, upon official notice of issuance, on NASDAQ.

5.13 Voting Trust. If at any time prior to the LCI Stockholders Meeting a third party shall make an unsolicited tender or exchange offer to acquire control of LCI, which offer is not recommended by LCI's Board of Directors, then Qwest and LCI will use their reasonable best efforts to consummate the transactions contemplated hereby by implementing a "voting trust" or similar structure permitting consummation of the transactions contemplated hereby prior to the receipt of final FCC and PUC approvals.

ARTICLE VI

CONDITIONS PRECEDENT

6.1 Conditions to Each Party's Obligation to Effect the Merger. The obligations of LCI, Qwest and Merger Sub to effect the Merger are subject to the satisfaction or waiver on or prior to the Closing Date of the following conditions:

(a) Stockholder Approval. (i) LCI shall have obtained the Required LCI Vote in connection with the adoption of this Agreement by the stockholders of LCI and (ii) Qwest shall have obtained the Required Qwest

Vote in connection with the approval of the Share Issuance by the stockholders of Qwest.

(b) No Injunctions or Restraints, Illegality. No Laws shall have been adopted or promulgated, and no temporary restraining order, preliminary or permanent injunction or other order issued by a court or other Governmental Entity of competent jurisdiction shall be in effect, having the effect of making the Merger illegal or otherwise prohibiting consummation of the Merger; provided, however, that the provisions of this Section 6.1(b) shall not be available to any party whose failure to fulfill its obligations pursuant to Section 5.4 shall have been the cause of, or shall have resulted in, such order or injunction.

(c) FCC and Public Utility Commission Approvals. All approvals for the Merger from the FCC and from the PUCs shall have been obtained other than those the failure of which to be obtained would not reasonably be expected to have individually or in the aggregate a Material Adverse Effect on Qwest and its Subsidiaries (including the Surviving Corporation and its Subsidiaries), taken together.

(d) HSR Act. The waiting period (and any extension thereof) applicable to the Merger under the HSR Act shall have been terminated or shall have expired.

(e) NASDAQ Listing. The shares of Qwest Common Stock to be issued in the Merger and such other shares to be reserved for issuance in connection with the Merger shall have been approved upon official notice of issuance for quotation on NASDAQ.

(f) Effectiveness of the Form S-4. The Form S-4 shall have been declared effective by the SEC under the Securities Act. No stop order suspending the effectiveness of the Form S-4 shall have been issued by the SEC and no proceedings for that purpose shall have been initiated or threatened by the SEC.

6.2 Additional Conditions to Obligations of Qwest and Merger Sub. The obligations of Qwest and Merger Sub to effect the Merger are subject to the satisfaction of, or waiver by Qwest, on or prior to the Closing Date of the following conditions:

(a) Representations and Warranties. Each of the representations and warranties of LCI set forth in this Agreement that is qualified as to materiality shall have been true and correct on the date of this Agreement, and each of the representations and warranties of LCI that is not so qualified shall have been true and correct in all material respects on the date of this Agreement, and Qwest shall have received a certificate of the chief executive officer and the chief financial officer of LCI to such effect.

(b) Performance of Obligations of LCI. LCI shall have performed or complied with all agreements and covenants required to be performed by it under this Agreement at or prior to the Closing Date that are qualified as to materiality and shall have performed or complied in all material respects with all other agreements and covenants required to be performed by it under this Agreement at or prior to the Closing Date that are not so qualified as to materiality, and Qwest shall have received a

certificate of the chief executive officer and the chief financial officer of LCI to such effect.

(c) Tax Opinion. Qwest shall have received from O'Melveny & Myers LLP, counsel to Qwest, on the Closing Date, a written opinion dated as of such date substantially in the form of Exhibit 6.2(c)(1). In rendering such opinion, counsel to Qwest shall be entitled to rely upon representations of officers of Qwest and LCI substantially in the form of Exhibits 6.2(c)(2) and 6.2(c)(3) (allowing for such amendments to the representations as counsel to Qwest deems necessary).

6.3 Additional Conditions to Obligations of LCI. The obligations of LCI to effect the Merger are subject to the satisfaction of, or waiver by LCI, on or prior to the Closing Date of the following additional conditions:

(a) Representations and Warranties. Each of the representations and warranties of Qwest and Merger Sub set forth in this Agreement that is qualified as to materiality shall have been true and correct on the date of this Agreement, and each of the representations and warranties of each of Qwest and Merger Sub that is not so qualified shall have been true and correct in all material respects on the date of this Agreement, and LCI shall have received a certificate of the chief executive officer and the chief financial officer of Qwest to such effect.

(b) Performance of Obligations of Qwest. Qwest shall have performed or complied with all agreements and covenants required to be performed by it under this Agreement at or prior to the Closing Date that are qualified as to materiality and shall have performed or complied in all material respects with all agreements and covenants required to be performed by it under this Agreement at or prior to the Closing Date that are not so qualified as to materiality, and LCI shall have received a certificate of the chief executive officer and the chief financial officer of Qwest to such effect.

(c) Tax Opinion. LCI shall have received from Kramer, Levin, Naftalis & Frankel, counsel to LCI, on the Closing Date, a written opinion dated as of such date substantially in the form of Exhibit 6.3(c)(1). In rendering such opinion, counsel to LCI shall be entitled to rely upon representations of officers of Qwest and LCI substantially in the form of Exhibits 6.2(c)(2) and 6.2(c)(3) (allowing for such amendments to the representations as counsel to LCI deems necessary).

ARTICLE VII

TERMINATION AND AMENDMENT

7.1 Termination. This Agreement may be terminated at any time prior to the Effective Time, by action taken or authorized by the Board of Directors of the terminating party or parties, and except as provided below, whether before or after approval of the matters presented in connection with the Merger by the stockholders of LCI or Qwest:

(a) By mutual written consent of Qwest and LCI, by action of their respective Boards of Directors;

(b) By either LCI or Qwest if the Effective Time shall not have occurred on or before the first anniversary of the date of this Agreement (the "Termination Date"); provided, however, that the right to terminate this Agreement under this Section 7.1(b) shall not be available to any party whose failure to fulfill any obligation under this Agreement (including without limitation Section 5.4) has to any extent been the cause of, or resulted in, the failure of the Effective Time to occur on or before the Termination Date;

By either LCI or Qwest if any Governmental Entity (i) (C) shall have issued an order, decree or ruling or taken any other action (which the parties shall have used their best efforts to resist, resolve or lift, as applicable, in accordance with Section 5.4) permanently restraining, enjoining or otherwise prohibiting the transactions contemplated by this Agreement, and such order, decree, ruling or other action shall have become final and nonappealable or (ii) shall have failed to issue an order, decree or ruling or to take any other action (which order, decree, ruling or other action the parties shall have used their best efforts to obtain, in accordance with Section 5.4), in each case (i) and (ii) which is necessary to fulfill the conditions set forth in subsections 6.1(c) and (d), as applicable, and such denial of a request to issue such order, decree, ruling or take such other action shall have become final and nonappealable; provided, however, that the right to terminate this Agreement under this Section 7.1(c) shall not be available to any party whose failure to comply with Section 5.4 has to any extent been the cause of such action or inaction;

(d) By either LCI or Qwest if (i) the approval by the stockholders of LCI required for the consummation of the Merger shall not have been obtained by reason of the failure to obtain the Required LCI Vote or (ii) the approval by the stockholders of Qwest required for the consummation of the Merger shall not have been obtained by reason of the failure to obtain the Required Qwest Vote, in each case upon the taking of such vote at a duly held meeting of stockholders of LCI or Qwest, as the case may be, or at any adjournment thereof;

(e) By Qwest if the Board of Directors of LCI, prior to the LCI Stockholders Meeting (i) shall withdraw or modify in any adverse manner the Board Approval, (ii) shall approve or recommend a Superior Proposal pursuant to Section 5.5 or (iii) shall resolve to take any of the actions specified in clauses (i) or (ii) above; (f) By LCI at any time prior to the LCI Stockholders Meeting, upon three Business Days' prior notice to Qwest, if the Board of Directors of LCI shall approve a Superior Proposal; provided, however, that (i) LCI shall have complied with Section 5.5, (ii) the Board of Directors of LCI shall have concluded in good faith, after giving effect to all concessions which may be offered by Qwest pursuant to clause (iii) below, on the basis of the advice of its financial advisors and outside counsel, that such proposal is a Superior Proposal and (iii) prior to any such termination, LCI shall, and shall cause its financial and legal advisors to, negotiate with Qwest to make such adjustments in the terms and conditions of this Agreement as would enable Qwest to proceed with the transactions contemplated hereby; provided, however, that it shall be a condition to termination by LCI pursuant to this Section 7.1(f) that LCI shall have made the payment of the Termination Fee to Qwest required by Section 7.2(b);

(g) By Qwest if any Person who is not an affiliate of Qwest shall have acquired more than 50% of the LCI Common Stock;

(h) By Qwest if a Stock Acquisition Date (as defined in the Rights Agreement) shall have occurred; and

(i) By LCI, if its Board of Directors so determines by a vote of the majority of the members of its entire Board, at any time during the three-Business Day period commencing on the Determination Date (the "LCI Evaluation Period"), if the Average Price is less than \$26.9531, subject, (A) if LCI elects to exercise its termination however, to the following: right pursuant to this Section 7.1(i), it shall give Qwest written notice of its intention to terminate (the "Termination Notice"), which termination shall be effective at the close of business on the third Business Day following the delivery of the Termination Notice (which Termination Notice may be withdrawn by LCI at any time prior to the effectiveness of such termination), (B) during the two-Business Day period commencing with the delivery of a Termination Notice (the "Adjustment Election Period"), Qwest shall have the option of adjusting the Exchange Ratio to equal the quotient determined by dividing \$42 by the Average Price (rounded to the nearest 1/10,000) by delivering written notice to LCI within such two-Business Day period of its intention to so adjust the Exchange Ratio and (C) if Qwest makes an election to adjust the Exchange Ratio pursuant to the preceding clause (B) (an "Adjustment Election"), then this Agreement shall not terminate pursuant to this Section 7.1(i) and this Agreement shall remain in effect in accordance with its terms (except as the Exchange Ratio shall have been so modified), and any references in this Agreement to "Exchange Ratio" shall thereafter be deemed to refer to the Exchange Ratio as adjusted pursuant to this Section 7.1(i).

7.2 Effect of Termination.

(a) In the event of termination of this Agreement by either LCI or Qwest as provided in Section 7.1, this Agreement shall forthwith become void and there shall be no liability or obligation on the part of Qwest or LCI or their respective officers or directors except with respect to Section 3.1(i), Section 3.2(h), the second sentence of Section 5.3, Section 5.7, this Section 7.2 and Article VIII.

Qwest and LCI agree that LCI shall pay to Qwest the sum of (b) \$133 million (the "Termination Fee") solely as follows: (i) if LCI shall terminate this Agreement pursuant to Section 7.1(f), (ii) if (A) LCI or Qwest shall terminate this Agreement pursuant to Section 7.1(d)(i) due to the failure of LCI's stockholders to approve and adopt this Agreement, (B) at any time after the date of this Agreement and at or before the time of the event giving rise to such termination there shall exist an Acquisition Proposal with respect to LCI and (C) within 12 months of the termination of this Agreement, LCI enters into a definitive agreement with any third party with respect to an Acquisition Proposal or an Acquisition Proposal is consummated, (iii) if Owest shall terminate this Agreement pursuant to Section 7.1(e), 7.1(g) or 7.1(h), or (iv) if (A) Qwest shall terminate this Agreement pursuant to Section 7.1(b) or LCI or Qwest shall terminate this Agreement pursuant to Section 7.1(c), (B) at any time after the date of this Agreement and at or before the time of the event giving rise to such termination there shall exist an Acquisition Proposal, (C) following the existence of such Acquisition Proposal and prior to any such termination, LCI shall have intentionally breached (and not cured after notice thereof) any of its

material covenants or agreements set forth in this Agreement in any material respect and (D) within 12 months of any such termination of this Agreement, LCI shall enter into a definitive agreement with any third party with respect to an Acquisition Proposal or an Acquisition Proposal is consummated.

(c) The Termination Fee required to be paid pursuant to Section 7.2(b) shall be made prior to, and shall be a pre-condition to the effectiveness of termination of this Agreement by LCI pursuant to Section 7.1(f). Any other payment required to be made pursuant to Section 7.2(b) shall be made to Qwest not later than two Business Days after the entering into of a definitive agreement with respect to, or the consummation of, an Acquisition Proposal, as applicable, or a termination pursuant to Section 7.1(e), 7.1(g) or 7.1(h). All payments under this Section 7.2 shall be made by wire transfer of immediately available funds to an account designated by the party entitled to receive payment.

7.3 Amendment. This Agreement may be amended by the parties hereto, by action taken or authorized by their respective Boards of Directors, at any time before or after approval of the matters presented in connection with the Merger by the stockholders of LCI and Qwest, but, after any such approval, no amendment shall be made which by law or in accordance with the rules of any relevant stock exchange requires further approval by such stockholders without such further approval. This Agreement may not be amended except by an instrument in writing signed on behalf of each of the parties hereto.

7.4 Extension; Waiver. At any time prior to the Effective Time, the parties hereto, by action taken or authorized by their respective Boards of Directors, may, to the extent legally allowed, (i) extend the time for the performance of any of the obligations or other acts of the other parties hereto, (ii) waive any inaccuracies in the representations and warranties contained herein or in any document delivered pursuant hereto and (iii) waive compliance with any of the agreements or conditions contained herein. Any agreement on the part of a party hereto to any such extension or waiver shall be valid only if set forth in a written instrument signed on behalf of such party. The failure of any party to this Agreement to assert any of its rights under this Agreement or otherwise shall not constitute a waiver of those rights.

ARTICLE VIII

GENERAL PROVISIONS

8.1 Non-Survival of Representations, Warranties and Agreements. None of the representations, warranties, covenants and other agreements in this Agreement or in any instrument delivered pursuant to this Agreement, including any rights arising out of any breach of such representations, warranties, covenants and other agreements, shall survive the Effective Time, except for those covenants and agreements contained herein and therein that by their terms apply or are to be performed in whole or in part after the Effective Time and this Article VIII. Nothing in this Section 8.1 shall relieve any party for any breach of any representation, warranty, covenant or other agreement in this Agreement occurring prior to termination.

8.2 Notices. All notices and other communications hereunder shall be in writing and shall be deemed duly given (a) on the date of delivery if delivered personally, or by telecopy or telefacsimile, upon confirmation of receipt, (b) on the first Business Day following the date of dispatch if delivered by a recognized next-day courier service, or (c) on the tenth Business Day following the date of mailing if delivered by registered or certified mail, return receipt requested, postage prepaid. All notices hereunder shall be delivered as set forth below, or pursuant to such other instructions as may be designated in writing by the party to receive such notice:

(a) if to Qwest or Merger Sub, to

Qwest Communications International Inc. 1000 Qwest Tower 555 Seventeenth Street Denver, Colorado 80202 Fax: (303) 291-1724 Attention: Robert S. Woodruff

with a copy to

O'Melveny & Myers LLP 153 East 53rd Street

New York, New York 10022-4611 Fax: (212) 326-2061 Attention: Drake S. Tempest and with a copy to Holme Roberts & Owen LLP 1700 Lincoln Street Denver, Colorado 80203 Fax: (303) 866-0200 Attention: Martha D. Rehm if to LCI to (b) LCI International, Inc. 8180 Greensboro Drive Suite 800 McLean, Virginia 22102 Fax: (703) 714-1750 Attention: General Counsel with a copy to Simpson Thacher & Bartlett 425 Lexington Avenue New York, New York 10017-3954 Fax: (212) 455-2502 Attention: Robert E. Spatt and with a copy to Kramer, Levin, Naftalis & Frankel 919 Third Avenue New York, New York 10022 Fax: (212) 715-8000 Attention: Peter S. Kolevzon

8.3 Interpretation. When a reference is made in this Agreement to Sections, Exhibits or Schedules, such reference shall be to a Section of or Exhibit or Schedule to this Agreement unless otherwise indicated. The table of contents, glossary of defined terms and headings contained in this Agreement are for reference purposes only and shall not affect in any way the meaning or interpretation of this Agreement. Whenever the words "include", "includes" or "including" are used in this Agreement, they shall be deemed to be followed by the words "without limitation".

8.4 Counterparts. This Agreement may be executed in one or more counterparts, all of which shall be considered one and the same

agreement and shall become effective when one or more counterparts have been signed by each of the parties and delivered to the other party, it being understood that both parties need not sign the same counterpart.

8.5 Entire Agreement; No Third Party Beneficiaries.

(a) This Agreement and the agreements referred to in Sections 1.9 and 5.3 constitute the entire agreement and supersede all prior agreements and understandings, both written and oral, among the parties with respect to the subject matter hereof, other than the Confidentiality Agreement, which shall survive the execution and delivery of this Agreement.

(b) This Agreement shall be binding upon and inure solely to the benefit of each party hereto, and nothing in this Agreement, express or implied, is intended to or shall confer upon any other Person any right, benefit or remedy of any nature whatsoever under or by reason of this Agreement, other than Section 5.8 (which is intended to be for the benefit of the Persons covered thereby and may be enforced by such Persons).

8.6 Governing Law. This Agreement shall be governed and construed in accordance with the laws of the State of Delaware.

8.7 Severability. If any term or other provision of this Agreement is invalid, illegal or incapable of being enforced by any law or public policy, all other terms and provisions of this Agreement shall nevertheless remain in full force and effect so long as the economic or legal substance of the transactions contemplated hereby is not affected in any manner materially adverse to any party. Upon such determination that any term or other provision is invalid, illegal or incapable of being enforced, the parties hereto shall negotiate in good faith to modify this Agreement so as to effect the original intent of the parties as closely as possible in an acceptable manner in order that the transactions contemplated hereby are consummated as originally contemplated to the greatest extent possible.

8.8 Assignment. Neither this Agreement nor any of the rights, interests or obligations hereunder shall be assigned by any of the parties hereto, in whole or in part (whether by operation of law or otherwise), without the prior written consent of the other party, and any attempt to make any such assignment without such consent shall be null and void, except that Merger Sub may assign, in its sole discretion, any or all of its rights,

interests and obligations under this Agreement to any direct wholly owned Subsidiary of Qwest without the consent of LCI, but no such assignment shall relieve Merger Sub of any of its obligations under this Agreement. Subject to the preceding sentence, this Agreement will be binding upon, inure to the benefit of and be enforceable by the parties and their respective successors and assigns.

8.9 Submission to Jurisdiction; Waivers. Each of Qwest and LCI irrevocably agrees that any legal action or proceeding with respect to this Agreement or for recognition and enforcement of any judgment in respect hereof brought by the other party hereto or its successors or assigns may be brought and determined in the Chancery or other Courts of the State of Delaware, and each of Qwest and LCI hereby irrevocably submits with regard to any such action or proceeding for itself and in respect to its property, generally and unconditionally, to the nonexclusive jurisdiction of the aforesaid courts. Each of Qwest and LCI hereby irrevocably waives, and agrees not to assert, by way of motion, as a defense, counterclaim or otherwise, in any action or proceeding with respect to this Agreement, (a) any claim that it is not personally subject to the jurisdiction of the above-named courts for any reason other than the failure to serve process in accordance with this Section 8.9, (b) that it or its property is exempt or immune from jurisdiction of any such court or from any legal process commenced in such courts (whether through service of notice, attachment prior to judgment, attachment in aid of execution of judgment, execution of judgment or otherwise), and (c) to the fullest extent permitted by applicable law, that (i) the suit, action or proceeding in any such court is brought in an inconvenient forum, (ii) the venue of such suit, action or proceeding is improper and (iii) this Agreement, or the subject matter hereof, may not be enforced in or by such courts. This Agreement does not involve less than \$100,000, and the parties intend that 6 Del.C. Section 2708 shall apply to this Agreement.

8.10 Enforcement. The parties agree that irreparable damage would occur in the event that any of the provisions of this Agreement were not performed in accordance with their specific terms. It is accordingly agreed that the parties shall be entitled to specific performance of the terms hereof, this being in addition to any other remedy to which they are entitled at law or in equity.

8.11 Definitions. As used in this Agreement:

(a) "Benefit Plans" means, with respect to any Person, each employee benefit plan, program, arrangement and contract (including, without limitation, any "employee benefit plan," as defined in Section 3(3) of the Employee Retirement Income Security Act of 1974, as amended ("ERISA") and any bonus, deferred compensation, stock bonus, stock purchase, restricted stock, stock option, employment, termination, stay agreement or bonus, change in control and severance plan, program, arrangement and contract) all of the foregoing in effect on the date of this Agreement or disclosed on Section 4.1(c) of the LCI Disclosure Schedule, to which such Person or its Subsidiary is a party, which is maintained or contributed to by such Person, or with respect to which such Person could incur material liability under Section 4069, 4201 or 4212(c) of ERISA.

(b) "Board of Directors" means the Board of Directors of any specified Person and any committees thereof.

(c) "Business Day" means any day on which banks are not required or authorized to close in the City of New York.

(d) "Material Adverse Effect" means, with respect to any

entity, any adverse change, circumstance or effect that, individually or in the aggregate with all other adverse changes, circumstances and effects, is or is reasonably likely to be materially adverse to the business, financial condition or results of operations of such entity and its Subsidiaries taken as a whole, other than any change, circumstance or effect relating to (i) the economy or securities markets in general or (ii) the industries in which Qwest or LCI operate and not specifically relating to Qwest or LCI.

(e) "the other party" means, with respect to LCI, Qwest and means, with respect to Qwest, LCI.

(f) "Person" means an individual, corporation, limited liability company, partnership, association, trust, unincorporated organization, other entity or group (as defined in the Exchange Act).

(g) "Subsidiary" when used with respect to any party means any corporation or other organization, whether incorporated or unincorporated, (i) of which such party or any other Subsidiary of such party is a general partner (excluding partnerships, the general partnership interests of which held by such party or any Subsidiary of such party do not have a majority of the voting interests in such partnership) or (ii) at least a majority of the securities or other interests of which having by their terms ordinary voting power to elect a majority of the Board of Directors or others performing similar functions with respect to such corporation or other organization is directly or indirectly owned or controlled by such party or by any one or more of its Subsidiaries, or by such party and one or more of its Subsidiaries.

"Superior Proposal" means a bona fide written Acquisition (h) Proposal which the Board of Directors of LCI concludes in good faith (after consultation with its financial advisors and legal counsel), taking into account all legal, financial, regulatory and other aspects of the proposal and the Person making the proposal, (i) would, if consummated, result in a transaction that is more favorable to LCI's stockholders (in their capacities as stockholders), from a financial point of view, than the transactions contemplated by this Agreement and (ii) is reasonably capable of being completed (provided that for purposes of this definition the term Acquisition Proposal shall have the meaning assigned to such term in Section 5.5 except that the references to "15%" and "20%" in the definition of "Acquisition Proposal" shall each be deemed to be a reference to "50%" and "Acquisition Proposal" shall only be deemed to refer to a transaction involving LCI, or with respect to assets (including the shares of any Subsidiary of LCI) of LCI and its Subsidiaries, taken as a whole, and not any of its Subsidiaries alone).

8.12 Other Agreements. The parties hereto acknowledge and agree that, except as otherwise expressly set forth in this Agreement, the rights and obligations of LCI and Qwest under any other agreement between the parties shall not be affected by any provision of this Agreement. IN WITNESS WHEREOF, Qwest, LCI and Merger Sub have caused this Agreement to be signed by their respective officers thereunto duly authorized, all as of March 8, 1998.

QWEST COMMUNICATIONS INTERNATONAL INC.

By: /s/ Joseph P. Nacchio Name:Joseph P. Nacchio Title:President

QWEST 1998-L ACQUISITION CORP.

By: /s/ Joseph P. Nacchio Name:Joseph P. Nacchio Title:President

LCI INTERNATIONAL, INC.

Exhibit 5.6

Assumption of LCI Stock Options and LCI Warrant

(1) Each LCI Stock Option and LCI Warrant outstanding at the Effective Time shall be assumed by Qwest and deemed to constitute an option to acquire, and each LCI Warrant shall be deemed to constitute a warrant to acquire, on the same terms and conditions, mutatis mutandis, as were applicable under such LCI Stock Option or such LCI Warrant prior to the Effective Time, the number of shares of Qwest Common Stock as the holder of such LCI Stock Option or LCI Warrant would have been entitled to receive pursuant to the Merger had such holder exercised such LCI Stock Option or such LCI Warrant in full immediately prior to the Effective Time (not taking into

account whether or not such option was in fact exercisable) at a price per share equal to (a) the aggregate exercise price for LCI Common Stock otherwise purchasable pursuant to such LCI Stock Option divided by (b) the number of shares of Qwest Common Stock deemed purchasable pursuant to such assumed LCI Stock Option or such LCI Warrant; provided that the number of shares of Qwest Common Stock that may be purchased upon exercise of any such LCI Stock Option or such LCI Warrant shall not include any fractional share and, upon exercise of such LCI Stock Option or such LCI Warrant, a cash payment shall be made for any fractional share based upon the last sale price per share of Qwest Common Stock on the trading day immediately preceding the date of exercise. All such LCI Stock Options shall be immediately exercisable by the holder thereof at or after the Effective Time, notwithstanding any provision to the contrary set forth in any option agreement (including, without limitation, any provision restricting the acceleration of such options in respect of any limit on deductibility under Section 280G of the Code). Within three Business Days after the Effective Time, Qwest shall cause to be delivered to each holder of an outstanding LCI Stock Option an appropriate notice setting forth such holder's rights pursuant thereto, and such assumed LCI Stock Option or such LCI Warrant (as adjusted with respect to exercise price and the number of shares of Qwest Common Stock purchasable) shall continue in effect on the same terms and conditions. From and after the Effective Time, Owest shall comply with the terms of the warrant agreement pursuant to which the LCI Warrant was issued (the "Warrant Agreements") and the LCI Stock Option Plans pursuant to which the LCI Stock Options were granted. The adjustments provided in this Exhibit 5.6 with respect to any Stock Options that are "incentive stock options" (as defined in Section 422 of the Code) shall be effected in a manner consistent with Section 424(a) of the Code.

(2) Qwest shall cause to be taken all corporate action necessary to reserve for issuance a sufficient number of shares of Qwest Common Stock for delivery upon exercise of LCI Stock Options and LCI Warrants in accordance with this Exhibit 5.6. Within three Business Days after the Effective Time, Qwest shall cause the Qwest Common Stock subject to LCI Stock Options and, to the extent required by the respective Warrant Agreements, subject to the LCI Warrants to be registered under the Securities Act pursuant to a registration statement on Form S-8 (or any successor or other appropriate forms), and shall use its best efforts to cause the effectiveness of such registration statement (and the current status of the prospectus or prospectuses contained therein) to be maintained for so long as the LCI Stock Options remain outstanding.

(3) LCI shall take such action as is necessary to cause the ending date of the then current offering period under each LCI employee stock purchase plan (including any stock purchase plan of a company acquired by LCI) to be prior to the Effective Time and on such date as is determined in accordance with the terms of such plan (the "Final Purchase Date"); provided that such change in the offering period shall be conditioned upon the consummation of the Merger. On the Final Purchase Date, the LCI shall apply the funds credited as of such date under such LCI Employee Stock Purchase Plan within each participant's payroll withholding account to the purchase of whole shares of LCI Common Stock in accordance with the terms of such LCI employee stock purchase plan.

(4) LCI shall use its best efforts to provide, on or prior to the Closing Date, a written acknowledgment of each holder of a LCI Stock Option or an LCI Warrant that such LCI Stock Option or LCI Warrant from and after the Effective Time is exercisable for shares of Qwest Common Stock as provided herein; provided that LCI need not do so if Qwest determines to its reasonable satisfaction that the terms of such LCI Stock Option or LCI Warrant provide that, after giving effect to any permitted action by the Board of Directors of LCI or any committee thereof, from and after the Effective Time, such LCI Stock Option or LCI Warrant shall be exercisable only for shares of Qwest Common Stock and not for shares for stock of the Surviving Corporation or any other entity. VOTING AGREEMENT dated as of March 8, 1998 (this "Agreement") among PHILIP F. ANSCHUTZ ("Shareholder"), ANSCHUTZ COMPANY, a Delaware corporation that is wholly owned by Shareholder ("Record Holder"), and LCI INTERNATIONAL, INC., a Delaware corporation (together with its successors and assigns, "LCI").

RECITALS

A. Shareholder beneficially owns not less than 170,000,000 shares of Common Stock, par value \$.01 per share, of Qwest Corporation, a Delaware corporation ("Qwest" and the "Qwest Common Stock"). All such shares, together with all other shares of capital stock of Qwest with respect to which Shareholder has beneficial ownership as of the date of this Agreement, are referred to as the "Subject Shares"; provided that any such share shall cease to be a "Subject Share" from and after the time that such share is transferred pursuant to Section 2 and ceases to be subject to the Voting Documents (as defined below) in accordance with the terms of Section 2. Record Holder is the record owner of all of the Subject Shares.

B. Qwest, Qwest 1998-L Acquisition Corp., a Delaware corporation ("Merger Sub"), and LCI are, simultaneously with the execution hereof, entering into an Agreement and Plan of Merger dated as of March 8, 1998 (the "Merger Agreement") providing for the merger of Merger Sub with and into LCI (the "Merger"). Terms not otherwise defined in this Agreement have the meanings stated in the Merger Agreement.

C. The Board of Directors of Qwest has approved an amendment to the Amended and Restated Certificate of Incorporation of Qwest to increase the number of authorized shares of Qwest Common Stock.

D. Shareholder, Record Holder and LCI desire to enter into this Agreement to provide for, among other things, (1) the obligation of Shareholder to cause Record Holder to vote the Subject Shares at the Qwest Stockholders Meeting to approve the Share Issuance and the Qwest Charter Amendment and (2) certain restrictions on (A) the sale or other transfer of the record ownership or the beneficial ownership, or both, of the Subject Shares by Shareholder and Record Holder and (B) the acquisition by Shareholder or Record Holder of beneficial ownership of additional shares of capital stock of Qwest from any Person other than Qwest, in each case until the consummation of the Merger or the termination of the Merger Agreement. This Agreement and all other agreements, instruments and other documents executed and delivered by Shareholder and Record Holder in connection with this Agreement are collectively referred to as the "Voting Documents".

E. Shareholder and Record Holder acknowledge that LCI is entering into the Merger Agreement in reliance on the representations, warranties,

covenants and other agreements of Shareholder and Record Holder set forth in this Agreement and would not enter into the Merger Agreement if Shareholder and Record Holder did not enter into this Agreement.

AGREEMENT

The parties agree as follows:

Section 1. Covenants of Shareholder and Record Holder.

(a) Voting. Until the day following the termination of this Agreement, subject to the receipt of proper notice and the absence of a preliminary or permanent injunction or other final order by any United States federal court or state court barring such action, Shareholder shall cause Record Holder to do, and Record Holder shall do, the following:

> (1) be present, in person or represented by proxy, at each meeting (whether annual or special, and whether or not an adjourned or postponed meeting) of the stockholders of Qwest, however called, or in connection with any written consent of the stockholders of Qwest, so that all Subject Shares then entitled to vote may be counted for the purposes of determining the presence of a quorum at such meetings; and

> (2) at each such meeting held before the Effective Time and with respect to each such written consent, vote (or cause to be voted), or deliver a written consent (or cause a consent to be delivered) covering, all the Subject Shares to approve the Share Issuance and the Qwest Charter Amendment and any action required in furtherance thereof.

(b) Stock Acquisitions. Until the day following the termination of this Agreement, Shareholder shall not, and shall cause Record Holder and his other affiliates not to, acquire, from any Person other than Qwest, beneficial ownership of any additional shares of Qwest Common Stock; provided, however, that Qwest may purchase shares of Qwest Common Stock to the extent permitted by the Merger Agreement and subject to the terms thereof.

(c) No Inconsistent Agreements. Until the day following the termination of this Agreement, Shareholder and Record Holder, shall not enter into any voting agreement or grant a proxy or power of attorney with respect to the Subject Shares which is inconsistent with this Agreement.

(d) Review of Merger Agreement. Both Shareholder and Record Holder acknowledge receipt and review of a copy of the Merger Agreement.

Section 2. Transfer of Subject Shares. During the term of this Agreement, Shareholder agrees not to transfer record or beneficial ownership of any shares of capital stock of Record Holder, and Shareholder and Record Holder shall not transfer record ownership or beneficial ownership, or both, of any Subject Shares except in each case to the extent permitted in the following sentences. Shareholder may transfer record or beneficial ownership, or both, of any shares of capital stock of Record Holder to any Person that is wholly owned, directly or indirectly, by Shareholder; provided that each such Person, and each Person that directly or indirectly is the record or beneficial owner of the shares of capital stock of such Person, shall then be a party to this Agreement or shall have executed and delivered an agreement by which such Person agrees to be bound by Sections 1, 2 and 4 of this Agreement with respect to such shares. Shareholder and Record Holder may transfer record ownership or beneficial

ownership, or both, of any Subject Shares, and such shares shall cease to be subject to the Voting Documents; provided, that (x) if, as a result of such transfer, less than 51% of the outstanding shares of Qwest Common Stock would be subject to the Voting Documents, then the Person to whom record ownership or beneficial ownership, or both, of such shares shall be transferred shall execute and deliver to LCI an agreement by which such transferee agrees that such shares shall be Subject Shares that are subject to the Voting Documents and agrees to be bound by Sections 1, 2 and 4 of this Agreement with respect to such shares and (y) in any event, on the record date for the meeting of the stockholders of Qwest at which the Share Issuance and the Qwest Charter Amendment shall be presented for their approval or with respect to any written consent in lieu thereof, the Subject Shares shall constitute a majority of the outstanding shares of Qwest Common Stock. For the purposes of this Agreement, the term "transfer" means a sale, an assignment, a grant, a transfer, a pledge, the creation of a lien or other disposition of any Subject Shares or any interest of any nature in any Subject Shares, including, without limitation, the "beneficial ownership" of such Subject Shares (as determined pursuant to Regulation 13D-G under the Exchange Act).

Section 3. Representations and Warranties of Shareholder. Shareholder and Record Holder, jointly and severally, represent and warrant to LCI as follows:

(a) Existence and Power. Record Holder (1) is a corporation duly incorporated, validly existing and in good standing under the laws of the State of Delaware and (2) has all requisite corporate power and authority to execute and deliver each Voting Document to which it is or may become a party.

(b) Authorization; Contravention. The execution and delivery by Shareholder and Record Holder of each Voting Document and the performance by it of its obligations under each Voting Document have, (1) in the case of Record Holder, been duly authorized by all necessary corporate action and (2) do not and will not conflict with or result in a Violation pursuant to, (A) in the case of Record Holder, any provision of its certificate of incorporation or bylaws, or (B) in the case of Shareholder and Record Holder, any loan or credit agreement, note, mortgage, bond, indenture, lease, benefit plan or other agreement, obligation, instrument, permit, concession, franchise, license, judgment, order, decree, statute, law, ordinance, rule or regulation applicable to Shareholder or Record Holder, as the case may be, the Subject Shares or any of Shareholder's or Record Holder's other properties or assets.

(c) Binding Effect. Each Voting Document constitutes, or when executed and delivered by Shareholder and Record Holder will constitute, a valid and binding obligation of Shareholder and Record Holder, respectively, enforceable against Shareholder or Record Holder, as the case may be, in accordance with its terms, except as such enforceability may be limited by bankruptcy, insolvency, reorganization, moratorium and similar laws relating to or affecting creditors generally, by general equity principles (regardless of whether such enforceability is considered in a proceeding in equity or at law) or by an implied covenant of good faith and fair dealing.

(d) Ownership. Shareholder and Record Holder are the only beneficial owners of the Subject Shares, free and clear of all liens, and Record Holder is the sole record holder of the Subject Shares, free and clear of liens, except in each case with respect to pledges or other liens that

Shareholder or Record Holder or both would be entitled to effect or create as of the date of this Agreement pursuant to the third sentence of Section 2 and in accordance with the terms thereof. As of the date of this Agreement, Shareholder does not own beneficially or of record any equity securities of Qwest other than the Subject Shares and Record Holder does not own beneficially or of record any equity securities of Qwest other than the Subject Shares and the Qwest Warrants. Shareholder is the sole record and beneficial owner of all of the capital stock of Record Holder, free and clear of all liens. Neither Shareholder nor Record Holder has appointed or granted any proxy which is still effective with respect to Subject Shares.

(e) Litigation. There is no action, suit, investigation, complaint or other proceeding pending against Shareholder or Record Holder or, to the knowledge of Shareholder or Record Holder, threatened against Shareholder or Record Holder or any other Person that restricts in any material respect or prohibits (or, if successful, would restrict or prohibit) the exercise by any party or beneficiary of its rights under any Voting Document or the performance by any party of its obligations under any Voting Document.

Section 4. Miscellaneous Provisions.

(a) Notices. All notices and other communications hereunder shall be in writing and shall be deemed duly given (1) on the date of delivery if delivered personally, or by telecopy or telefacsimile, upon confirmation of receipt, (2) on the first Business Day following the date of dispatch if delivered by a recognized next-day courier service, or (3) on the tenth Business Day following the date of mailing if delivered by registered or certified mail, return receipt requested, postage prepaid. All notices hereunder shall be given the party at its address stated on the signature pages of this Agreement or at any other address as the party may specify for this purpose by notice to the other party. (b) No Waivers; Remedies; Specific Performance.

(1) No failure or delay by LCI in exercising any right, power or privilege under any Voting Document shall operate as a waiver of the right, power or privilege. A single or partial exercise of any right, power or privilege shall not preclude any other or further exercise of the right, power or privilege or the exercise of any other right, power or privilege. The rights and remedies provided in the Voting Documents shall be cumulative and not exclusive of any rights or remedies provided by law.

(2) In view of the uniqueness of the agreements contained in the Voting Documents and the transactions contemplated hereby and thereby and the fact that LCI would not have an adequate remedy at law for money damages in the event that any obligation under any Voting Document is not performed in accordance with its terms, each of Shareholder and Record Holder therefore agrees that LCI shall be entitled to specific enforcement of the terms of each Voting Document in addition to any other remedy to which LCI may be entitled, at law or in equity.

(c) Amendments, Etc. No amendment, modification, termination, or waiver of any provision of any Voting Document, and no consent to any departure by any of Shareholder, Record Holder and LCI from any provision of

any Voting Document, shall be effective unless it shall be in writing and signed and delivered by each of Shareholder, Record Holder and LCI, and then it shall be effective only in the specific instance and for the specific purpose for which it is given.

(d) Successors and Assigns; Third Party Beneficiaries.

(1) No party shall assign any of its rights or delegate any of its obligations under any Voting Document. Any assignment or delegation in contravention of this Section 4(d) shall be void ab initio and shall not relieve the assigning or delegating party of any obligation under any Voting Document.

(2) The provisions of each Voting Document shall be binding upon and inure solely to the benefit of the parties hereto, the express beneficiaries thereof (to the extent provided therein) and their respective permitted heirs, executors, legal representatives, successors and assigns, and no other person.

(e) Governing Law. Each Voting Document and all rights, remedies, liabilities, powers and duties of the parties hereto and thereto, shall be governed in accordance with the laws of the State of Delaware without regard to principles of conflicts of laws.

(f) Severability of Provisions. If any term or other provision of any Voting Document is invalid, illegal or incapable of being enforced by any

law or public policy, all other terms and provisions of such Voting Document shall nevertheless remain in full force and effect so long as the economic or legal substance of the transactions contemplated hereby is not affected in any manner materially adverse to any party. Upon such determination that any term or other provision is invalid, illegal or incapable of being enforced, the parties shall negotiate in good faith to modify such Voting Document so as to effect the original intent of the parties as closely as possible in an acceptable manner in order that the transactions contemplated hereby are consummated as originally contemplated to the greatest extent possible.

(q) Headings and References. Article and section headings in any Voting Document are included for the convenience of reference only and do not constitute a part of the Voting Document for any other purpose. References to parties, express beneficiaries, articles and sections in any Voting Document are references to parties to or the express beneficiaries and sections of the Voting Document, as the case may be, unless the context shall require otherwise. Any of the terms defined in this Agreement may, unless the context otherwise requires, be used in the singular or the plural, depending on the reference. The use in this Agreement of the word "include" or "including," when following any general statement, term or matter, shall not be construed to limit such statement, term or matter to the specific items or matters set forth immediately following such word or to similar items or matters, whether or not nonlimiting language (such as "without limitation" or "but not limited to" or words of similar import) is used with reference thereto, but rather shall be deemed to refer to all other items or matters that fall within the broadest possible scope of such general statement, term or matter.

(h) Entire Agreement. The Voting Documents embody the entire agreement and understanding of Shareholder, Record Holder and LCI, and

supersede all prior agreements or understandings, with respect to the subject matters of the Voting Documents.

(i) Survival. Except as otherwise specifically provided in any Voting Document, each representation, warranty or covenant of a party contained in the Voting Document shall remain in full force and effect, notwithstanding any investigation or notice to the contrary or any waiver by any other party or beneficiary of a related condition precedent to the performance by the other party or beneficiary of an obligation under the Voting Document.

(j) Submission to Jurisdiction; Waivers. Each of Shareholder, Record Holder and LCI irrevocably agrees that any legal action or proceeding with respect to any Voting Document or for recognition and enforcement of any judgment in respect hereof or thereof brought by the other party hereto or its successors or assigns may be brought and determined in the Chancery or other Courts of the State of Delaware, and each of Shareholder, Record Holder and LCI hereby irrevocably submits with regard to any such action or proceeding for itself and in respect to its property, generally and unconditionally, to the nonexclusive jurisdiction of the aforesaid courts. Each of Shareholder, Record Holder and LCI hereby irrevocably waives, and agrees not to assert, by way of motion, as a defense, counterclaim or otherwise, in any action or proceeding with respect to any Voting Document, (a) any claim that it is not personally subject to the jurisdiction of the above-named courts for any reason other than the failure to serve process in accordance with this Section 4(j), (b) that it or its property is exempt or immune from jurisdiction of any such court or from any legal process commenced in such courts (whether through service of notice, attachment prior to judgment, attachment in aid of execution of judgment, execution of judgment or otherwise), and (c) to the fullest extent permitted by applicable law, that (i) the suit, action or proceeding in any such court is brought in an inconvenient forum, (ii) the venue of such suit, action or proceeding is improper and (iii) such Voting Document, or the subject matter hereof or thereof, may not be enforced in or by such courts. This Agreement does not involve less than \$100,000, and the parties intend that 6 Del.C. Section 2708 shall apply to this Agreement.

(k) Waiver of Jury Trial. Each party, and each express beneficiary of a Voting Document as a condition of its right to enforce or defend any right under or in connection with such Voting Document, waives any right to a trial by jury in any Action to enforce or defend any right under any Voting Document and agrees that any Action shall be tried before a court and not before a jury.

(1) Termination. LCI may terminate this Agreement at any time upon written notice to each of Shareholder and Record Holder. Unless terminated earlier by LCI or by mutual agreement of the parties, this Agreement shall terminate upon the first to occur of (i) consummation of the Merger and (ii) the termination of Merger Agreement pursuant to Section 7.1 thereof.

(m) Counterparts. This Agreement may be signed in any number of counterparts, each of which shall be an original, with the same effect as if all signatures were on the same instrument.

IN WITNESS WHEREOF, the parties have executed and delivered this Agreement as of the date first written above.

PHILIP F. ANSCHUTZ

/s/ Philip F. Anschutz

Address: 2400 Qwest Tower 555 Seventeenth Street Denver, Colorado 80202

Telecopy: (303) 298-8881

ANSCHUTZ COMPANY

- By: /s/ Cannon Y. Harvey Name: Cannon Y. Harvey Title: President
- Address: 2400 Qwest Tower 555 Seventeenth Street Denver, Colorado 80202
- Telecopy: (303) 298-8881

LCI INTERNATIONAL, INC.

By: /s/ H. Brian Thompson Name: H. Brian Thompson Title: Chairman of the Board and Chief Executive Officer Address: 8180 Greensboro Drive Suite 800 McLean, Virginia 22102

Telecopy: (703) 714-1750

FIRST AMENDMENT to RIGHTS AGREEMENT

FIRST AMENDMENT dated as of March 8, 1998 ("this Amendment") between LCI International, Inc., a Delaware corporation (the "Company"), and Fifth Third Bank, as Rights Agent.

WHEREAS, the above-mentioned parties have previously entered into that certain Rights Agreement dated as of January 22, 1997 (the "Rights Agreement") governing certain preferred stock purchase rights (the "Rights") of the Company's stockholders;

WHEREAS, the Company has entered into an Agreement and Plan of Merger (the "Merger Agreement"), by and among the Company, Qwest Communications International Inc., a Delaware corporation ("Parent"), and Quest 1998-L Acquisition Corp, a Delaware corporation and a wholly-owned subsidiary of Parent ("QAI"), pursuant to which QAI will be merged with and into the Company and the Company shall become a wholly-owned subsidiary of Parent (the "Merger");

WHEREAS, the Board of Directors deems it desirable and in the best interests of its stockholders that the transactions contemplated by the Merger Agreement be consummated;

WHEREAS, the Merger Agreement provides that prior to the execution of the Merger Agreement, the Board of Directors of the Company shall have approved an amendment to the Rights Agreement to the effect that Parent, QAI and their affiliates shall not become an Acquiring Person (as such term is defined in the Rights Agreement); and

WHEREAS, such parties wish to amend the Rights Agreement in the manner set forth below.

NOW, THEREFORE, the parties hereto agree as follows:

 All capitalized terms used herein, unless otherwise defined herein, shall have the meanings given them in the Rights Agreement, and each reference in the Rights Agreement to "this Agreement," "hereof," "herein," "hereunder" or "hereby" and each other similar reference shall be deemed to refer to the Rights Agreement as amended hereby. All references to the Rights Agreement in any other agreement between or among any of the parties hereto relating to the transactions contemplated by the Rights Agreement shall be deemed to refer to the Rights Agreement as amended hereby.

2. The definition of "Acquiring Person" in Section 1 of the Rights Agreement is hereby amended by adding the following provision to the end of the last sentence of paragraph (a)(i) of such definition: "and shall not include Parent, QAI or any of their Affiliates or Associates which otherwise would become an Acquiring Person solely by reason or as a result of the execution or delivery of the Merger Agreement or the consummation of the Merger or any other transaction contemplated by the Merger Agreement."

3. The following definitions are hereby added to Section 1 of the Rights Agreement:

""Merger" means the merger of QAI with and into the Company upon the terms and conditions set forth in the Merger Agreement."

""Merger Agreement" means the Agreement and Plan of Merger, dated as of March 8, among the Company, QAI and Parent."

""Parent" means Qwest Communications International Inc., a Delaware corporation."

""QAI" means Qwest 1998-L Acquisition Corp., a Delaware corporation and a direct, wholly owned subsidiary of Parent."

4. This Amendment shall be governed by and construed in accordance with the internal laws of the State of Delaware applicable to contracts made and performed entirely in Delaware.

5. This Amendment may be signed in any number of counterparts, each of which shall be deemed an original, with the same effect as if the signatures thereto and hereto were upon the same instrument.

6. Except as expressly amended hereby, the Rights Agreement shall remain in full force and effect.

IN WITNESS WHEREOF, the parties hereto have caused this Amendment to be duly executed and their respective corporate seals to be hereunto affixed and attested, all as of the day and year first above written.

LCI INTERNATIONAL, INC.

Attest: /s/ James D. Heflinger

/s/ H. Brian Thompson By:-----

> Name: H. Brian Thompson Title: Chairman

FIFTH THIRD BANK

Attest: /s/ Laura H. Wikoff

Secretary

/s/ Dana S. Hushak

Name: Dana S. Hushak Title: Vice President For Immediate Release

Qwest Media Contact: Alexander Communications	Qwest Investor Contact: Qwest Communications
Erin McKelvey	Lee Wolfe
(303) 615-5070 ext. 108	(800) 567-7296
emckelve@alexander-pr.com	1wolfe@qwest.net
	http://www.qwest.net
LCI Media Contact:	LCI Investor Contact:
LCI International	LCI International

LCI International	LCI International
Gerry Simone	Kevin Taback
(703) 848-4478	(703) 848-4493
simoneg@lci.com	tabackk@lci.com
http://www.lci.com	http://www.lci.com

QWEST AND LCI INTERNATIONAL ANNOUNCE \$4.4 BILLION MERGER Combined Entity To Become Fourth Largest Long-Distance Company

Denver, CO/McLean, VA - March 9, 1998 - Qwest Communications International Inc. and LCI International, Inc. today announced the signing of a definitive merger agreement. The merger will create the fourth-largest U.S. long distance company and one of the fastest growing companies in the communications industry. The all-stock transaction is valued at approximately \$4.4 billion. The merger enables the LCI nationwide customer base to fully leverage the capabilities and efficiencies of the Qwest Macro Capacity SM fiber network and allows Qwest to take full advantage of LCI's sales and marketing expertise, distribution channels, intelligent network platform, as well as the company's customer care and billing system. The combined companies had 1997 revenues of \$2.3 billion, serve over two million business and residential customers and have a total current equity market capitalization of over \$11 billion. (See attached fact sheet for further detail.)

The board of directors of each company have approved the merger. The terms of the merger agreement call for each LCI share to be converted into \$42.00 of Qwest common stock, under current market conditions. Based on the closing prices of Qwest and LCI on March 6, 1998, LCI shareholders would receive approximately 122.4 million newly issued shares of Qwest stock, or 36.4 percent of the combined company's shares. The merger is intended to qualify as a tax-free reorganization and will be accounted for as a purchase.

"Brian Thompson and the LCI team have built a world class communications company. Combining their customer base and their highly competitive people with those at Qwest, and their state of the art billing and customer care platforms with the Qwest network creates a powerful new company," said Joseph P. Nacchio, president and CEO of Qwest. "Qwest's ability to sell next generation data services is greatly enhanced by the LCI sales, marketing and distribution strengths. This merger accelerates both Qwest's and LCI's business plans by several years."

The combination of Qwest and LCI will result in significant operating and financial benefits. The merger will deliver greater network efficiencies, eliminate duplicate efforts to build sales and systems infrastructure, avoid duplication of capital spending programs and accelerate the companies' data and international strategies. "LCI has achieved a 40 percent average annual revenue growth rate over the last 5 years and is one of the most successful and fastest growing telecommunications carriers in the U.S.," said H. Brian Thompson, chairman and CEO of LCI. "We anticipate that Qwest's technologically advanced network will enable us to further accelerate our performance."

An integration team composed of four senior executives from each company has been formed to oversee the combination process to ensure that operational benefits are fully realized. LCI will receive two seats on Qwest's board of directors, one of which will be filled by H. Brian Thompson, who will become vice chairman of the board. Completion of the transaction is anticipated to occur during the third quarter of 1998. The transaction is subject to the majority vote of the shareholders of Qwest and LCI and to other customary conditions such as receipt of regulatory approvals. The majority shareholder of Qwest has agreed to vote in favor of the transaction.

The actual number of shares of Qwest common stock to be exchanged for each LCI share will be determined by dividing \$42 by a 15-day volume weighted average of trading prices for Qwest common stock prior to the closing, but will not be less than 1.0625 shares (if Qwest's average stock price exceeds \$39.53) or more than 1.5583 shares (if Qwest's average stock price is less than \$26.95). If Qwest's average stock price is less than \$26.95, LCI may terminate the merger unless Qwest then agrees to exchange for each share of LCI the number of Qwest shares determined by dividing \$42 by such average price.

The Qwest Macro Capacity Fiber Network

Qwest's planned domestic 16,285 mile network will serve more than 125 cities, which represent approximately 80% of the data and voice traffic originating in the United States, upon its scheduled completion in the second quarter of 1999. Currently, more than 3,600 miles are activated from Los Angeles to Columbus, Ohio. Qwest is also extending its network 1,400 miles into Mexico with completion slated for late third quarter 1998.

The Qwest Macro Capacity Fiber network is designed with a highly reliable and secure bi-directional, line switching OC-192 SONET ring architecture. Upon completion, the network will offer a self-healing system that provides the ultimate security and reliability by allowing instantaneous rerouting in the event of a fiber cut or equipment failure.

About LCI International

LCI International, Inc. (NYSE:LCI), one of the nation's fastest-growing long-distance telecommunications carriers, provides a full array of worldwide voice and data transmission services to businesses, residential customers and other carriers through its 4,500 mile fiber-optic network is planned to be 8,500 miles by the end of the year. LCI International, Inc. is headquartered in McLean, VA, with offices in more than 60 locations, including national network control and customer service centers, and regional operations in various locations throughout the United States.

About Qwest

Qwest Communications International Inc. (NASDAQ:QWST) is a multimedia communications company building a high-capacity, fiber optic network for the 21st century. With its cutting-edge technology, Qwest will deliver high-quality data, video and voice connectivity securely and reliably to businesses, consumers and other communications service providers. Further information is available at www.qwest.net.

Lehman Brothers Inc. acted as financial advisor to LCI and Salomon Smith Barney acted as financial advisor to Qwest.

This release, including the attached merger fact sheet, may contain forward-looking statements that involve risks and uncertainties. These statements may differ materially from actual future events or results. Readers are referred to the documents filed by Qwest with the SEC, specifically its most recent reports on Form 10-Q, which identify important risk factors that could cause actual results to differ from those contained in the forward-looking statements, including potential fluctuations in quarterly results, dependence on new product development, delays in closing the transaction, inability to achieve combination efficiencies, rapid technological and market change, failure to complete the network on schedule, volatility of stock price, financial risk management and future growth subject to risks.

This announcement is not an offer to sell or a solicitation to buy any securities of Qwest. The offering with respect to the proposed merger with LCI will be made only by the proxy statement/prospectus that will be distributed to stockholders of LCI in connection with their consideration of the transaction.

 ${\tt QwestLinked}$ is a trademark, and the Qwest logo is a registered trademark of Qwest Communications International Inc. in the U.S. and certain other countries.

###

Qwest Communications

LCI International

Merger Fact Sheet

Creates fourth largest long-distance company, as measured by revenues. (assuming Worldcom and MCI merger is consummated.)

Combined market capitalization exceeds \$11 billion, at March 6th trading prices.

More fully utilizes Qwest's 16,000 mile Macro Capacity sm fiber network.

Combined employee base will be approximately 5,800 (1,800 Qwest and 4,000 LCI).

Combined customer base will be approximately 2.3 million.

Combines the industry's most seasoned and entrepreneurial management teams.

The combined companies will operate in over 70 U.S. locations (10 Qwest, 60+ LCI), including major operations in Denver, Colorado, Northern Virginia, Columbus, Ohio and San Antonio, Texas. Qwest's headquarters will remain in Denver, CO. Operating synergies will be significant in all areas of operations. Revenues Network Facility Costs Sales and Marketing Network Engineering and Operations Information Technology Other Administration and Corporate Total cost synergies for the first full year post-combination are estimated to exceed \$300 million and exceed \$1 billion for 1998 through 2001. Capital expenditure synergies are estimated to be approximately \$80 million in 1999 and exceed \$285 million for 1998 through 2001. Presentation includes forward-looking statements. Actual results may vary. Key 1997 Operating Data: <TABLE> <CAPTION> Qwest LCI Combined _____ _____ _____ <S> <C> <C> <C> Revenues \$115.3 \$1,642.0 \$1,757.3 Communications services \$581.4 \$0.0 Construction services \$581.4 \$1,642.0 \$696.7 \$2,338.7 Total Revenues Revenue Growth ('97 v. '96) 202% 26% Ebitda<F1> \$115.2 \$194.0 \$309.2 Ebitda Growth<F1> ('97 v. '96) 507% 29% Net Income<F2> \$14.5 \$97.0 \$111.5 Net Income Growth<F2> 308% 30% ('97 v. '96) \$1,390.0 Total Assets \$1,354.0 \$2,744.0 669.2 12,903.0 Minutes of Use 13,572.2 Minutes of Use Growth 75% 35% ('97 v. '96) _____ <FN> <F1>Adjusted for Qwest Growth Shares <F2>Adjusted for LCI non-recurring charges </TABLE>

Post Acquisition Stock Ownership:

36.4% LCI pro forma ownership of combined entity based on March 6, 1998 closing stock price.